

**CSA Notice of Approval** Amendments to National Instrument 23-101 Trading Rules and Companion Policy 23-101CP to National Instrument 23-101 Trading Rules

April 7, 2016

### Introduction

The Canadian Securities Administrators (the CSA or we) have approved amendments to National Instrument 23-101 Trading Rules (the Instrument, or NI 23-101) and its related Companion Policy (23-101CP) (together, the Amendments) and are finalizing a methodology for the regulatory oversight of market data fees (the Data Fees Methodology).

We are publishing the text of the Amendments in Annex B to this notice, together with certain other relevant information at Annexes C through G. The text of the Amendments will also be available on the websites of the CSA jurisdictions, including:

www.lautorite.qc.ca www.albertasecurities.com www.bcsc.bc.ca www.gov.ns.ca/nssc www.fcnb.ca www.osc.gov.on.ca www.fcaa.gov.sk.ca www.msc.gov.mb.ca

Provided all necessary ministerial approvals are obtained, the Amendments will come into force on July 6, 2016, except as related to the market share threshold. The Amendments related to the market share threshold will come into force on October 1, 2016.

### **Substance and Purpose**

The substance and purpose of the Amendments is to update NI 23-101 and 23-101CP in relation to the application of the order protection rule (OPR), and in response to recent market developments. The Amendments adjust the rule framework in a manner that maintains the core principles of OPR, but address some of the inefficiencies and costs that have resulted from its implementation. Further, the Amendments add OPR-related guidance to 23-101CP to address circumstances where a marketplace has introduced an intentional order processing delay.

### Background

Following a CSA review of OPR, we published proposed amendments to NI 23-101 and 23-101CP on May 15, 2014 (the 2014 Notice).<sup>1</sup> After considering the comments received in response to the initial publication, we have made non-material changes to certain aspects of the proposals. For additional background on the substance and purpose of the proposed amendments, please refer to the 2014 Notice.

In addition, on June 12, 2015 we published proposed amendments to 23-101CP to address changes introduced to the functionality of certain marketplaces (the 2015 Notice).<sup>2</sup> These marketplace functionality changes impose delays on the entry of orders into the trading engine of the marketplace that would, for a period of time, prevent the immediate execution of orders that are submitted to execute against displayed volume. The introduction of these delays, referred to as 'speed bumps', raised the question of whether OPR should apply to displayed orders on such marketplaces. In response to comments received to the 2015 Notice, we have made non-material changes to the language originally proposed. For additional background, please refer to the 2015 Notice.

### Summary of Written Comments Received by the CSA

In response to the 2014 Notice and the 2015 Notice, we received submissions from 27 commenters and 14 commenters, respectively. We have considered the comments received and thank all of the commenters for their input. A list of those who submitted comments and a summary of the comments and our responses are attached at Annexes D and E to this notice. Copies of the comment letters are available at www.osc.gov.on.ca.

### Summary of the Amendments and Notable Changes

See Annex A for a summary of the Amendments, and a description of notable changes that have been made to the amendments proposed in the 2014 Notice and the 2015 Notice.

### **Local Matters**

Certain jurisdictions are publishing other information required by local securities legislation. In Ontario, this information is contained in Annex G of this notice.

### Annexes

- A. Summary of the Amendments and Notable Changes;
- B. Amendments to NI 23-101;
- C. Blackline showing changes to NI 23-101 and 23-101CP;

<sup>&</sup>lt;sup>1</sup> Published on May 15, 2014, at: (2014) 37 OSCB 4873.

<sup>&</sup>lt;sup>2</sup> Published on June 12, 2015, at: (2015) 38 OSCB 5551.

- D. List of commenters who provided submissions on the proposed amendments published on May 15, 2014 and the proposed amendments published on June 12, 2015;
- E. Summary of comments received on the 2014 Notice and the 2015 Notice, together with the CSA's response to the comments;
- F. Data Fees Methodology; and
- G. Local Matters.

# Questions

Please refer your questions to any of the following:

Tracey Stern	Timothy Baikie
Manager, Market Regulation	Senior Legal Counsel, Market Regulation
Ontario Securities Commission	Ontario Securities Commission
tstern@osc.gov.on.ca	tbaikie@osc.gov.on.ca
Alina Bazayan	Louis-Philippe Pellegrini
	11 0
Senior Analyst, Market Regulation	Legal Counsel, Market Regulation
Ontario Securities Commission	Ontario Securities Commission
abazavan@osc.gov.on.ca	lpellegrini@osc.gov.on.ca
Paul Redman	Serge Boisvert
Principal Economist, Strategy & Operations	Senior Policy Advisor
Ontario Securities Commission	Direction des bourses et des OAR
predman@osc.gov.on.ca	Autorité des marchés financiers
	serge.boisvert@lautorite.qc.ca
Roland Geiling	Lynn Tsutsumi
Derivatives Product Analyst	Director, Market Regulation
Direction des bourses et des OAR	Alberta Securities Commission
Autorité des marchés financiers	lynn.tsutsumi@asc.ca
roland.geiling@lautorite.qc.ca	
Sasha Cekerevac	Bruce Sinclair
Regulatory Analyst, Market Regulation	Securities Market Specialist
Alberta Securities Commission	British Columbia Securities Commission
sasha.cekerevac@asc.ca	bsinclair@bcsc.bc.ca

# ANNEX A

### Summary of the Amendments and Notable Changes

This Annex summarizes the Amendments and describes the notable changes from the proposed amendments published in both the 2014 Notice and the 2015 Notice. Further, it provides details regarding additional matters relating to the Amendments. The information in this annex is set out in the following sections:

1. Market Share Threshold

2. Locked or Crossed Orders

3. Trading Fees

4. Intentional Order Processing Delays

5. Data Fees Methodology

6. Best Execution Obligations and Disclosure

7. Pilot Study on Prohibition on Payment of Rebates by Marketplaces

For purposes of this Annex, we refer to marketplaces that will display orders that are protected pursuant to OPR, as 'protected marketplaces', and other marketplaces not displaying protected orders as 'unprotected marketplaces'.

### 1. Market Share Threshold

We continue to believe that OPR is a valuable part of the regulatory framework, but recognize that the current application of OPR has introduced inefficiencies and costs. Further, we believe the rule acts as a form of regulatory support for marketplaces by requiring that marketplace participants access either directly or indirectly, all protected marketplaces, and pay associated costs in doing so. The Amendments will provide flexibility to marketplace participants in determining if and when to access trading on certain marketplaces by limiting the application of OPR to orders displayed on marketplaces that meet a market share threshold determined by the CSA.

The comments received in relation to the publication of the 2014 Notice provided mixed views with respect to the market share threshold. Supporters of the threshold approach were not unanimous in support of the specific level of the threshold proposed, with arguments presented for both higher and lower percentages. Many commenters expressed the view that any threshold percentage applied would be an arbitrary figure. Those who were not supportive of the threshold approach conveyed concerns about both the impact on competition, as well as additional market complexities that would result from an environment where some visible marketplaces would

display orders that would be protected under OPR, while others would display orders that would not be protected.

We recognize the concerns expressed in some of the comment letters. However, after considering all of the comments received, we believe that at this time, the threshold approach is the most appropriate method of balancing the benefits of OPR with the costs associated with its application.

We remain supportive of marketplace competition, but believe that the benefits of competition should be achieved in combination with an allowance for users of marketplace services to exercise some element of discretion when determining whether to access marketplaces and pay for marketplace services. This is especially true in relation to the launch of new visible marketplaces in Canada, to which market participants are currently required to immediately connect or access in order to ensure compliance with OPR.

The market share threshold will be initially set in each jurisdiction by the regulator, or in Quebec, the securities regulatory authority, at 2.5% based on an average share of the adjusted<sup>3</sup> volume and value traded (equally weighted) over a one-year period,<sup>4</sup> and applied at the market or facility level where the marketplace is comprised of more than one visible market or facility.<sup>5</sup> Excluded from the market share threshold calculation will be:

- trades involving dark passive orders,
- the non-interfered portion of intentional crosses,<sup>6</sup>
- trades from call markets or call facilities (including existing opening and closing call facilities)
- odd-lot trades,
- auto-executed trades in fulfillment of market maker obligations or participation rights, and
- trades involving special terms orders.

The displayed orders of a recognized exchange that does not meet the market share threshold will be protected, but only with respect to those securities listed by and traded on the exchange. In these circumstances, in a similar manner to the application of the market share threshold, protection for displayed orders for listed securities on a recognized exchange will be applied at the market or facility level where the recognized exchange is comprised of more than one market or facility.

# (i) Market Share Threshold Calculation

<sup>&</sup>lt;sup>3</sup> Volume and value traded will be adjusted to exclude certain trades.

<sup>&</sup>lt;sup>4</sup> Volume and value will be calculated on a total market basis, rather than calculated separately on the basis of listing marketplace.

<sup>&</sup>lt;sup>5</sup> Certain marketplaces have distinct visible continuous auction order books, to which the market share threshold will be applied separately.

<sup>&</sup>lt;sup>6</sup> On some marketplaces, the execution of an intentional cross by a dealer can be broken up or "interfered" with by an existing order from same dealer, which has already been entered on the marketplace at the same price as the intentional cross. Because the interfering order would have been protected under OPR, it would be included in the market share calculation.

Once the threshold is set in each jurisdiction by the regulator, or in Quebec, the securities regulatory authority, the market share and the list of protected marketplaces will be made publicly available on the websites of both the CSA and the Investment Industry Regulatory Organization of Canada (IIROC). The initial calculation of protected marketplaces will be effective for a six month period and will be published during the first week of June 2016. It will come into effect on October 1, 2016 and will be based on trading data from June 1, 2015 to May 31, 2016.

After the initial period, on an ongoing basis, the application of OPR for displayed orders as it relates to the market share threshold, will be effective for a period of one year, and subject to annual renewal. Marketplace participants will be given approximately three months after each list is published to facilitate any required operational changes. The criteria for calculating the threshold and the process for communicating the list of protected marketplaces for the effective period will also be made publicly available, and any changes to the market share threshold will be set by the regulator, or in Quebec, the securities regulatory authority and will be communicated publicly via CSA Staff Notice.

For ease of reference, the following table outlines the initial 6 month timeframe and reference period, as well as the timing of the annual calculations thereafter.

	Calculation Period	Date of Publication of Protected Marketplace List	Effective Period
Initial Implementation	Trading data from June 1, 2015 to May 31, 2016	First week of June 2016	October 1, 2016 to March 31, 2017
First Full Implementation	Trading data from January 1, 2016 to December 31, 2016	By January 15, 2017	April 1, 2017
Ongoing Implementation	Trading data from the first through last day of the year annually	Annually by January 15 of each year	Annually on April 1

(ii) Ongoing Review of the Impact of the Threshold

We intend to monitor the impact of the threshold on an ongoing basis. We commit to conducting a review of the impact and cost savings associated with the market share threshold and the percentage at which it is set, once one year of data is available and can be analyzed.

(iii) Changes to Proposed Amendments in 2014 Notice

In the 2014 Notice we proposed to set the market share threshold at 5%. As noted above, the comments received in relation to the proposed threshold were mixed. As a result of both feedback received during the public comment process and CSA discussions, we intend to set the

market share threshold at 2.5% based on an equally weighted average of the share of adjusted volume and value traded.

This lower threshold will serve to address some of the concerns related to potential impacts on competition, but will still provide a base level of trading activity at or above which displayed orders will be protected. The lower market share threshold will still offer dealers an element of choice with respect to new marketplaces and those marketplaces below the threshold level.

# 2. Locked or Crossed Orders

The provisions in section 6.5 of the Instrument related to locked or crossed orders will be limited in application to "protected orders". This change will not preclude participants from entering orders on protected marketplaces that would lock or cross an order on an unprotected marketplace.

# 3. Trading Fees

The Amendments will introduce a cap on active trading fees charged by marketplaces.<sup>7</sup> As proposed in the 2014 Notice, the cap will apply to continuous auction trading in equity securities and exchange-traded funds (ETFs) and will be set at \$0.0030 per share or unit traded for securities priced at or above \$1.00, and \$0.0004 per share or unit traded for securities priced below \$1.00.

As was discussed in the 2014 Notice, the \$0.0030 per share cap for securities priced at or above \$1.00 is set at the same level as the cap set in the U.S. under Rule 610(c) of Regulation National Market System. We recognize that the trading fee cap is higher than the fees currently charged by most Canadian marketplaces and acknowledge feedback received as part of the public comment process that the cap is too high. We are finalizing the cap as proposed, in order to establish an interim ceiling on active trading fees while we consider additional steps to address the level of trading fees in Canada.

We recognize the views of some stakeholders that the fee cap should be lower. However, our market is highly integrated with the U.S., and there is significant trading activity in securities that are listed in both Canada and the U.S. (Inter-listed Securities). As a result, we are concerned about potential negative consequences for the Canadian market from establishing a trading fee cap for Inter-listed Securities that is significantly different than comparable regulatory requirements in the U.S. As liquidity providers are sensitive to rebates they receive for posting orders on certain marketplaces, a decrease in fees charged by those marketplaces would also result in a decrease in rebates available to liquidity providers. If the difference in rebates between Canada and the U.S. for Inter-listed Securities is too large, a shift of liquidity to U.S marketplaces and widening spreads on Canadian marketplaces could result.

However, in addition to the fee caps approved as part of the Amendments, we have published today a separate notice requesting comment on a revised active trading fee cap applicable only to

<sup>&</sup>lt;sup>7</sup> In the context of the Amendments, active trading fees refer to fees charged by marketplaces for the execution of an order that was entered to execute against a displayed order on that marketplace in continuous auction trading.

securities priced at or above \$1.00 that are listed on a Canadian exchange, but not also listed on a U.S. exchange (Non-Inter-listed Securities). The proposed cap on Non-Inter-listed Securities priced at or above \$1.00 would be \$0.0017 per share (further details can be found in that notice). Upon approval of this lower cap for Non-Inter-listed Securities, the active-trading fee cap of \$0.0030 per share for securities valued at \$1.00 or more would only apply to Inter-listed Securities.

# 4. Intentional Order Processing Delays

As previously noted, the 2015 Notice proposed amendments to 23-101CP related to the implementation of marketplace 'speed bumps' that delay the entry of orders into a marketplace trading engine. We proposed to add OPR-related guidance to 23-101CP, such that where a marketplace has introduced an intentional order processing delay that results in the inability to provide for an immediate execution against displayed volume, orders displayed on that marketplace would not be "protected orders" as defined in the Instrument.

After considering the comments received, we are finalizing those proposals with certain nonmaterial changes in response to requests by a number of commenters to clarify the language of the amendment. These changes are designed to provide greater clarity around how we interpret the definition of "automated trading functionality" in the Instrument and the types of factors considered in determining whether a marketplace offers the ability for an "immediate" execution.

# 5. Data Fees Methodology

To provide for a transparent process for regulatory oversight of real-time professional market data fees, we are finalizing and formally adopting the Data Fees Methodology proposed in the 2014 Notice, and currently being used informally in the review of professional market data fees in Ontario.

As discussed in the 2014 Notice, the Data Fees Methodology estimates a fee or fee range for topof-book (Level 1) and depth-of-book (Level 2) market data for each marketplace based on their contribution to price discovery and trading activity.

The Data Fees Methodology has a three step approach that involves:

- the calculation of pre- and post-trade metrics;
- a ranking of marketplaces on a relative basis; and
- an estimation of a fee or fee range for the professional market data fees charged by each marketplace based on a reference amount.

Where relevant to the calculations, the pre-trade metrics will include quotes displayed across all Canadian marketplaces whether these orders are considered protected or unprotected for the purposes of OPR.

The Data Fees Methodology will be used to assess the relative value of real-time market data feeds provided by each marketplace to its professional data subscribers, and will be applied in the context of:

- (a) an annual review of professional market data fees charged by each marketplace for both Level 1 and Level 2 data feeds and reapproval where fees are determined to be unreasonably high; and
- (b) the review and approval of any changes to Level 1 and Level 2 professional market data fees proposed by marketplaces.

Subsection 3.2(5) of National Instrument 21-101 *Marketplace Operation* requires each recognized exchange and alternative trading system (ATS) to file an updated and consolidated Form NI 21-101F1 or Form NI 21-101F2 within 30 days after the end of each calendar year. In Ontario, the OSC will apply the Data Fees Methodology to the Level 1 and Level 2 professional market data fees submitted in that consolidation under Exhibit L - Fees, to determine if the marketplace's fees are higher than the range identified through the Data Fees Methodology.

The Data Fees Methodology will apply to all marketplaces regardless of their protected or unprotected status. This is because we believe it is appropriate to maintain a level of oversight and ensure a consistent balance across all marketplaces, between the value assessed using the Data Fees Methodology and the associated fees that are charged for data. This is particularly important in the context of compliance with applicable best execution requirements.

(i) Changes to the Proposed Data Fee Methodology

We note that although the general approach to market data fees has not changed relative to that proposed in the 2014 Notice, to reflect stakeholders' comments and our ongoing observations, the Data Fees Methodology has been adjusted as discussed below. For more detailed information, please refer to Appendix A-2 of the 2014 Notice, and Annex F to this notice.

# (a) Pre- and Post- trade Metrics

Appendix A-2 to the 2014 Notice detailed a number of specific metrics, both pre- and post-trade which would be used to rank the relative contribution of each marketplace to price discovery and trading activity. One such metric, referred to as "\$Time(equal)", would measure a marketplace's contribution to the depth of liquidity quoted at the best bid and offer. Concerns have been expressed by stakeholders that the use of this specific metric in a ranking model would inflate the ranking of marketplaces that display quotes in illiquid securities and could be subject to manipulation. Given these concerns, this specific pre-trade metric and the ranking model that uses it (discussed below) will no longer be included in the Data Fees Methodology.

One proposed post-trade metric, referred to as "Scope of Trading," measured the number of symbols traded on each marketplace. In the 2014 Notice we identified certain disadvantages or downsides to the use of this metric in a ranking model, specifically that it might disproportionately advantage marketplaces with significant market share and disadvantage

smaller competitors or new entrants. The use of the metric could also "double penalize" marketplaces that do not trade all securities, and could potentially be manipulated. Given further consideration, we have excluded the "Scope of Trading" metric from the ranking model in which it is used (discussed below). However, we will continue to use this metric independently of the ranking model to better understand the range of securities traded on each marketplace compared to all securities traded across all marketplaces.

# (b) Ranking Models

Appendix A-2 to the 2014 Notice set out the proposed relative ranking models for marketplaces. Concerns were raised by commenters with respect to one ranking model, referred to as the "SIP(equal),<sup>8</sup>" that uses the "\$Time(equal)" metric discussed above. Specifically, commenters indicated that this ranking model would weigh the various pre-trade metrics it uses equally, rather than on the basis of value traded. Further, this ranking model would not distinguish between stocks that trade often and those that rarely trade, and as a result, would inflate the ranking of those marketplaces that trade in illiquid securities. Based on the concerns identified, we have excluded this ranking model from the Data Fees Methodology.

Further to the ranking models and as discussed above, in order to ensure that all marketplaces are treated fairly in the relative ranking process, we have adjusted the formula used for one specific ranking model, referred to as "Model 3," to specifically exclude the "Scope of Trading" metric that we initially proposed to use in its calculation.

# (c) Interim Reference Benchmark

In the 2014 Notice, we also highlighted two potential references (domestic and international) that could be used to allocate an estimated fee or fee range to a marketplace. We noted at that time that selecting the appropriate reference was a key element in ensuring the appropriate application of the Data Fee Methodology.

It is our intention to retain external assistance to determine the appropriate benchmark in the coming months. In the interim, we will apply the *domestic reference*, as described in the 2014 Notice, which aggregates the market data fees charged by all marketplaces into a single "pool" and then redistributes the amount based on the ranking models obtained through the Data Fee Methodology.

We recognize the concerns raised that existing fees are too high and a domestic benchmark based on an aggregated amount may be unreasonable. Despite this, we are of the view that we need to formally implement the Data Fee Methodology in order to manage existing fee levels. While the methodology will be used to address fees that are determined to be unreasonably high, we will not apply the methodology or the benchmark to support fee increases until such time as the appropriate benchmark has been established.

# (d) Non-professional Market Data User Fees

<sup>&</sup>lt;sup>8</sup> The SIP(equal) model is based on metrics used by the U.S. Securities Information Processor.

We indicated in the 2014 Notice that we were concerned about the level of real-time market data fees charged by marketplaces to non-professional data users in Canada. We also indicated that we were considering either a cap on non-professional data fees at a rate set as a percentage of that marketplace's reviewed and/or approved professional data subscriber rate, or the development of a methodology similar to the one applied to professional data users.

We will continue to monitor any developments in relation to the market data fees charged to nonprofessional users and will consider whether any action is necessary in the future.

# 6. Best Execution Obligations and Disclosure

We are finalizing changes to 23-101CP that will introduce guidance designed to provide greater clarity for dealers with respect to best execution and accessing marketplaces that a dealer is not required to access for purposes of regulatory compliance.

Further related to best execution, in the 2014 Notice we proposed amendments that would introduce new disclosure requirements for dealers regarding their best execution policies. This disclosure relates primarily to order handling / routing and potential conflicts of interest, and would ensure that clients are provided with a minimum level of information to assist in making informed decisions regarding the use of a dealer's services.

We are not finalizing the proposed amendments in relation to dealer best execution disclosure at this time. On December 10, 2015, IIROC published for comment *Proposed Provisions Respecting Best Execution* and related guidance<sup>9</sup>. The proposed IIROC amendments would update and consolidate best execution requirements in both the Universal Market Integrity Rules (UMIR) and Dealer Member Rules (DMR) into a single Dealer Member Rule respecting best execution. The updates as proposed would serve to assist dealers in complying with best execution obligations in a multiple marketplace environment, and would reflect the amendments proposed in the 2014 Notice by the CSA.

We believe that delaying the finalization of the amendments proposed in the 2014 Notice related to dealer best execution disclosure will allow us to benefit from any comments received in relation to the proposed IIROC amendments.

# 7. Pilot Study on Prohibition on Payment of Rebates by Marketplaces

In the 2014 Notice, we expressed our intention to move forward with a pilot study that would examine the impact of disallowing the payment of rebates by Canadian marketplaces. We stated our view that the payment of rebates by a marketplace, or any other entity, impacts behaviours of marketplace participants in ways which may be contributing to increased fragmentation and segmentation of order flow, distorting the rationale for investment or trading decisions, and creating unnecessary conflicts of interest for dealer routing decisions that may be difficult to manage.

<sup>&</sup>lt;sup>9</sup> Published at: http://www.iiroc.ca/Documents/2015/8df7a02c-4491-4fd0-b317-4c90bdc722a2\_en.pdf

Although we continue to believe that a pilot study would be useful to determine what, if any, impact would result from disallowing the payment of rebates by marketplaces, significant issues have been raised both in the context of the comment process as well as through follow-on meetings with both industry participants and academics. The primary issue relates to the inclusion of Inter-listed Securities in a pilot study, and the potential negative consequences if a similar pilot was not also implemented in the U.S. We share the concerns raised about the potential loss or migration of liquidity in Inter-listed Securities if we were to proceed absent similar regulatory requirements in the U.S. We considered operating a pilot study that would exclude Inter-listed Securities but given their significance in terms of volume and value of trading activity<sup>10</sup>, we are not certain that such a study would provide meaningful results. We will consider the possibility of a joint pilot if and when such an opportunity arises.

<sup>&</sup>lt;sup>10</sup> Based on 2015 data, we estimate that trading in Inter-listed Securities in Canada represented approximately 28% of total volume and approximately 56% of total value (source: Bloomberg).

### ANNEX B

### Amendments to National Instrument 23-101 Trading Rules

### 1. National Instrument 23-101 Trading Rules is amended by this Instrument.

### 2. Section 1.1 is amended by

(a) *replacing* "automated functionality" *with* "automated trading functionality" *in the definition of* "automated functionality",

### (b) replacing the definition of "directed-action order" with:

"directed-action order" means an order for the purchase or sale of an exchangetraded security, other than an option, that,

- (a) when entered on or routed to a marketplace, is to be immediately
  - (i) executed against a displayed order with any remainder to be booked or cancelled; or
  - (ii) placed in an order book;
- (b) is marked as a directed-action order; and
- (c) is entered on or routed to a marketplace
  - (i) to execute against a best-priced displayed order, or
  - (ii) at the same time that another order is entered on or routed to a marketplace to execute against any protected order with a better price than the entered or routed order;,
- (c) deleting "and" after "quoted;" in the definition of "non-standard order",
- (d) replacing paragraph (a) in the definition of "protected bid" with:
  - (a) that is displayed on a marketplace that provides automated trading functionality and
    - (i) the marketplace meets or exceeds the market share threshold as set for the purposes of this definition by the regulator, or in Québec, the securities regulatory authority; or

(ii) if the marketplace is a recognized exchange, the bid is for a security listed by and traded on that recognized exchange; and, *and* 

### (e) replacing paragraph (a) in the definition of "protected offer" with:

- (a) that is displayed on a marketplace that provides automated trading functionality and
  - the marketplace meets or exceeds the market share threshold as set for the purposes of this definition by the regulator, or in Québec, the securities regulatory authority; or
  - (ii) if the marketplace is a recognized exchange, the offer is for a security listed by and traded on that recognized exchange; and.
- 3. Subsection 6.3(2) is amended by replacing "a marketplace that routes an order to another marketplace must immediately notify" with "the marketplace that is executing the transaction or routing the order for execution must immediately notify the following of the failure, malfunction or material delay:".
- 4. *Subsection 6.3(3) is amended by adding* "displaying a protected order" *after* "concludes that a marketplace".
- 5. Subparagraph 6.4(1)(a)(ii) is amended by adding ";" after "traded through".
- 6. Section 6.5 is replaced with:

**6.5** Locked or Crossed Orders – A marketplace participant or a marketplace that routes or reprices orders must not intentionally enter a displayed order on a marketplace that is subject to section 7.1 of NI 21-101, at a price that,

- (a) in the case of an order to purchase, is the same as or higher than the best protected offer; or
- (b) in the case of an order to sell, is the same as or lower than the best protected bid.

### 7. The following section is added after section 6.6

### 6.6.1 Trading Fees

- (1) In this section, "exchange-traded fund" means a mutual fund,
  - (a) the units of which are listed securities or quoted securities, and

- (b) that is in continuous distribution in accordance with applicable securities legislation.
- (2) A marketplace that is subject to section 7.1 of NI 21-101 must not charge a fee for executing an order that was entered to execute against a displayed order on that marketplace greater than
  - (a) \$0.0030 per security traded for an equity security, or per unit traded for an exchange-traded fund, if the execution price of each security or unit traded is greater than or equal to \$1.00; or
  - (b) \$0.0004 per security traded for an equity security, or per unit traded for an exchange-traded fund, if the execution price of each security or unit traded is less than \$1.00.
- 8. Section 6.7 is amended by replacing "better-priced orders on a marketplace" with "better-priced protected orders".

# **Coming into force**

9.

- (1) Subject to subsection (2), this Instrument comes into force on July 6, 2016.
- (2) Paragraphs 2(d) and (e) of this Instrument come into force on October 1, 2016.

### ANNEX C

### NATIONAL INSTRUMENT 23-101

# TRADING RULES [Blacklined to current version]

### **Table of Contents**

Table of Contents PART TITLE			
PART 1	DEFINITION AND INTERPRETATION		
1.1 1.2	Definition Interpretation - NI 21-101		
1.2			
PART 2	APPLICATION OF THIS INSTRUMENT		
2.1	Application of this Instrument		
PART 3	MANIPULATION AND FRAUD		
3.1	Manipulation and Fraud		
PART 4	BEST EXECUTION		
4.1 4.2	Application of this Part Best Execution		
4.3	Order and Trade Information		
1.0			
PART 5	REGULATORY HALTS		
5.1	Regulatory Halts		
PART 6	ORDER PROTECTION		
6.1	Marketplace Requirements for Order Protection		
6.2	List of Trade-throughs		
6.3	Systems or Equipment Failure, Malfunction or Material Delay		
6.4	Marketplace Participant Requirements for Order Protection		
6.5	Locked or Crossed Orders		
6.6 <u>6.6</u> .1	Trading Hours Trading Fees		
<u>6.7</u>	Anti-Avoidance		
6.8	Application of this Part		
PART 7	MONITORING AND ENFORCEMENT OF REQUIREMENTS SET BY A RECOGNIZED		
	EXCHANGE AND A RECOGNIZED QUOTATION AND TRADE REPORTING SYSTEM		
7.1	Requirements for a Recognized Exchange		
7.2	Agreement between a Recognized Exchange and a Regulation Services Provider		
7.2.1	Obligations of a Recognized Exchange to a Regulation Services Provider		
7.3 7.4	Requirements for a Recognized Quotation and Trade Reporting System Agreement between a Recognized Quotation and Trade Reporting System and a Regulation		
7.4	Services Provider		
7.4.1	Obligations of a Quotation and Trade Reporting System to a Regulation Services Provider		
7.5	Co-ordination of Monitoring and Enforcement		
PART 8	MONITORING AND ENFORCEMENT REQUIREMENTS FOR AN ATS		
8.1	Pre-condition to Trading on an ATS		
8.2	Requirements Set by a Regulation Services Provider for an ATS		
8.3	Agreement between an ATS and a Regulation Services Provider		
8.4	Agreement between an ATS and its Subscriber		
8.5	[Repealed]		
PART 9	MONITORING AND ENFORCEMENT REQUIREMENTS FOR AN INTER-DEALER BOND		
	BROKER		
9.1	Requirements Set by a Regulation Services Provider for an Inter-Dealer Bond Broker		
9.2	Agreement between an Inter-Dealer Bond Broker and a Regulation Services Provider		

- 9.3 Exemption for an Inter-Dealer Bond Broker
- PART 10 MONITORING AND ENFORCEMENT REQUIREMENTS FOR A DEALER EXECUTING TRADES OF UNLISTED DEBT SECURITIES OUTSIDE OF A MARKETPLACE
- 10.1 Requirements Set by a Regulation Services Provider for a Dealer Executing Trades of Unlisted Debt Securities Outside of a Marketplace
- 10.2 Agreement between a Dealer Executing Trades of Unlisted Debt Securities Outside of a Marketplace and a Regulation Services Provider
- 10.3 [Repealed]
- PART 11 AUDIT TRAIL REQUIREMENTS
- 11.1 Application of this Part
- 11.2 Audit Trail Requirements for Dealers and Inter-Dealer Bond Brokers
- 11.3 Transmission in Electronic Form
- PART 12 EXEMPTION 12.1 Exemption
- PART 13 EFFECTIVE DATE 13.1 Effective Date

#### NATIONAL INSTRUMENT 23-101

#### TRADING RULES

#### PART 1 DEFINITION AND INTERPRETATION

#### **1.1 Definition** - In this Instrument

"automated trading functionality" means the ability to

- (a) immediately allow an incoming order that has been entered on the marketplace electronically to be marked as immediate-or-cancel;
- (b) immediately and automatically execute an order marked as immediate-or-cancel against the displayed volume;
- (c) immediately and automatically cancel any unexecuted portion of an order marked as immediate-or-cancel without routing the order elsewhere;
- (d) immediately and automatically transmit a response to the sender of an order marked as immediate-or-cancel indicating the action taken with respect to the order; and
- (e) immediately and automatically display information that updates the displayed orders on the marketplace to reflect any change to their material terms;

"best execution" means the most advantageous execution terms reasonably available under the circumstances;

"calculated-price order" means an order for the purchase or sale of an exchange-traded security, other than an option, that is entered on a marketplace and for which the price of the security

- (a) is not known at the time of order entry; and
- (b) is not based, directly or indirectly, on the quoted price of an exchange-traded security at the time the commitment to execute the order was made;

"closing-price order" means an order for the purchase or sale of an exchange-traded security, other than an option, that is

- (a) entered on a marketplace on a trading day; and
- (b) subject to the conditions that
  - (i) the order be executed at the closing sale price of that security on the marketplace for that trading day; and
  - (ii) the order be executed subsequent to the establishment of the closing price;

"directed-action order" means a limitan order for the purchase or sale of an exchange-traded security, other than an option, that,

- (a) when entered on or routed to a marketplace, is to be immediately
  - (i) executed against a protected displayed order with any remainder to be booked or cancelled; or
  - (ii) placed in an order book;
- (b) is marked as a directed-action order; and
- (c) is entered <u>on</u> or routed to a marketplace
  - (i) to execute against a best-priced displayed order, or

- (ii) at the same time as one or more additional that another limit orders is that are entered on or routed to one or more a marketplaces, as necessary, to execute against any protected order with a better price than the ontered or routed order referred to in paragraph (a);
- "NI 21-101" means National Instrument 21-101 Marketplace Operation;

"non-standard order" means an order for the purchase or sale of an exchange-traded security, other than an option, that is entered on a marketplace and is subject to non-standardized terms or conditions related to settlement that have not been set by the marketplace on which the security is listed or quoted; and

"protected bid" means a bid for an exchange-traded security, other than an option,

- (a) that is displayed on a marketplace that provided automated <u>trading</u> functionality; and
  - (i) the marketplace meets or exceeds the market share threshold as set for the purposes of this definition by the regulator, or in Quebec, the securities regulatory authority; or
  - (ii) if the marketplace is a recognized exchange, the bid is for a security listed by and traded on that recognized exchange; and
- (b) about which information is required to be provided pursuant to Part 7 of NI 21-101 to an information processor or, if there is no information processor, to an information vendor that meets the standards set by a regulation services provider;

"protected offer" means an offer for an exchange-traded security, other than an option,

- (a) that is displayed on a marketplace that provides automated <u>trading</u> functionality; and
  - (i) the marketplace meets or exceeds the market share threshold as set for the purposes of this definition by the regulator, or in Quebec, the securities regulatory authority; or
  - (ii) if the marketplace is a recognized exchange, the offer is for a security listed by and traded on that recognized exchange; and
- (b) about which information is required to be provided pursuant to Part 7 of NI 21-101 to an information processor or, if there is no information processor, to an information vendor that meets the standards set by a regulation services provider;

"protected order" means a protected bid or protected offer; and

"trade-through" means the execution of an order at a price that is,

- (a) in the case of a purchase, higher than any protected offer, or
- (b) in the case of a sale, lower than any protected bid.

**1.2** Interpretation - NI 21-101 - Terms defined or interpreted in NI 21-101 and used in this Instrument have the respective meanings ascribed to them in NI 21-101.

#### PART 2 APPLICATION OF THIS INSTRUMENT

**2.1 Application of this Instrument -** A person or company is exempt from subsection 3.1(1) and Parts 4 and 5 if the person or company complies with similar requirements established by

- (a) a recognized exchange that monitors and enforces the requirements set under subsection 7.1(1) directly;
- (b) a recognized quotation and trade reporting system that monitors and enforces requirements set under subsection 7.3(1) directly; or
- (c) a regulation services provider.

#### PART 3 MANIPULATION AND FRAUD

#### 3.1 Manipulation and Fraud

(1) A person or company must not, directly or indirectly, engage in, or participate in any transaction or series of transactions, or method of trading relating to a trade in or acquisition of a security or any act, practice or course of conduct, if the person or company knows, or ought reasonably to know, that the transaction or series of transactions, or method of trading or act, practice or course of conduct

- (a) results in or contributes to a misleading appearance of trading activity in, or an artificial price for, a security or a derivative of that security; or
- (b) perpetrates a fraud on any person or company.

(2) In Alberta, British Columbia, Ontario, Québec and Saskatchewan, instead of subsection (1), the provisions of the *Securities Act* (Alberta), the *Securities Act* (British Columbia), the *Securities Act* (Ontario), the *Securities Act* and the *Derivatives Act* (Québec) and *The Securities Act*, 1988 (Saskatchewan), respectively, relating to manipulation and fraud apply.

#### PART 4 BEST EXECUTION

**4.1 Application of this Part -** This Part does not apply to a dealer that is carrying on business as an ATS in compliance with section 6.1 of NI 21-101.

**4.2 Best Execution -** A dealer and an adviser must make reasonable efforts to achieve best execution when acting for a client.

**4.3** Order and Trade Information - To satisfy the requirements in section 4.2, a dealer or adviser must make reasonable efforts to use facilities providing information regarding orders and trades.

### PART 5 REGULATORY HALTS

**5.1 Regulatory Halts -** If a regulation services provider, a recognized exchange, recognized quotation and trade reporting system or an exchange or quotation and trade reporting system that has been recognized for the purposes of this Instrument and NI 21-101 makes a decision to prohibit trading in a particular security for a regulatory purpose, a person or company must not execute a trade for the purchase or sale of that security during the period in which the prohibition is in place.

#### PART 6 ORDER PROTECTION

**6.1 Marketplace Requirements for Order Protection -** (1) A marketplace must establish, maintain and ensure compliance with written policies and procedures that are reasonably designed

- (a) to prevent trade-throughs on that marketplace other than the trade-throughs referred to in section 6.2; and
- (b) to ensure that the marketplace, when executing a transaction that results in a trade-through referred to in section 6.2, is doing so in compliance with this Part.

(2) A marketplace must regularly review and monitor the effectiveness of the policies and procedures required under subsection (1) and must promptly remedy any deficiencies in those policies and procedures.

(3) At least 45 days before implementation, a marketplace must file with the securities regulatory authority and, if applicable, its regulation services provider the policies and procedures, and any significant changes to those policies and procedures established under subsection (1).

- 6.2 List of Trade-throughs For the purposes of paragraph 6.1(1)(a) the permitted trade-throughs are
- (a) a trade-through that occurs when the marketplace has reasonably concluded that the marketplace displaying the protected order that was traded through was experiencing a failure, malfunction or material delay of its systems or equipment or ability to disseminate marketplace data;

- (b) the execution of a directed-action order;
- (c) a trade-through by a marketplace that simultaneously routes a directed-action order to execute against the total displayed volume of any protected order that is traded through;
- (d) a trade-through if, immediately before the trade-through, the marketplace displaying the protected order that is traded through displays as its best price a protected order with a price that is equal or inferior to the price of the trade-through;
- (e) a trade-through that results when executing
  - (i) a non-standard order;
  - (ii) a calculated-price order; or
  - (iii) a closing-price order;
- (f) a trade-through that was executed at a time when the best protected bid for the security traded through was higher than the best protected offer.

**6.3** Systems or Equipment Failure, Malfunction or Material Delay - (1) If a marketplace experiences a failure, malfunction or material delay of its systems, equipment or its ability to disseminate marketplace data, the marketplace must immediately notify

- (a) all other marketplaces;
- (b) all regulation services providers;
- (c) its marketplace participants; and
- (d) any information processor or, if there is no information processor, any information vendor that disseminates its data under Part 7 of NI 21-101.

(2) If executing a transaction described in paragraph 6.2(a), and a notification has not been sent under subsection (1), <u>a-the</u> marketplace that routes an order to another marketplace securing the transaction or routing the order for execution must immediately notify the following of the failure, malfunction or material delay:

- (a) the marketplace that it reasonably concluded is experiencing a failure, malfunction or material delay of its systems or equipment or its ability to disseminate marketplace data;
- (b) all regulation services providers;
- (c) its marketplace participants; and
- (d) any information processor disseminating information under Part 7 of NI 21-101.

(3) If a marketplace participant reasonably concludes that a marketplace <u>displaying a protected order</u> is experiencing a failure, malfunction or material delay of its systems or equipment or its ability to disseminate marketplace data, and routes an order to execute against a protected order on another marketplace displaying an inferior price, the marketplace participant must notify the following of the failure, malfunction or material delay

- (a) the marketplace that may be experiencing a failure, malfunction or material delay of its systems or equipment or its ability to disseminate marketplace data; and
- (b) all regulation services providers.

**6.4 Marketplace Participant Requirements for Order Protection -** (1) A marketplace participant must not enter a directed-action order unless the marketplace participant has established, and maintains and ensures compliance with, written policies and procedures that are reasonably designed

(a) to prevent trade-throughs other than the trade-throughs listed below:

- a trade-through that occurs when the marketplace participant has reasonably concluded that the marketplace displaying the protected order that was traded through was experiencing a failure, malfunction or material delay of its systems or equipment or ability to disseminate marketplace data;
- (ii) a trade-through by a marketplace participant that simultaneously routes a directed-action order to execute against the total displayed volume of any protected order that is traded through;
- a trade-through if, immediately before the trade-through, the marketplace displaying the protected order that is traded through displays as its best price a protected order with a price that is equal or inferior to the price of the trade-through transaction;
- (iv) a trade-through that results when executing
  - (A) a non-standard order;
  - (B) a calculated-price order; or
  - (C) a closing-price order;
- (v) a trade-through that was executed at a time when the best protected bid for the security traded through was higher than the best protected offer; and
- (b) to ensure that when executing a trade-through listed in paragraphs (a)(i) to (a)(v), it is doing so in compliance with this Part.

(2) A marketplace participant that enters a directed-action order must regularly review and monitor the effectiveness of the policies and procedures required under subsection (1) and must promptly remedy any deficiencies in those policies and procedures.

6.5 Locked or Crossed Orders - A marketplace participant or a marketplace that routes or reprices orders must not intentionally enter a displayed order on a marketplace that is subject to section 7.1 of NI 21-101, at a price that,

- (a) enter on a marketplace a protected order to buy a security at a price that<u>in the case of an order to purchase</u>, is the same as or higher than the best protected offer; or
- (b) enter on a marketplace a protected order to sell a security at a price that<u>in the case of an order to sell</u>, is the same as or lower than the best protected bid.
- 6.6 **Trading Hours -** A marketplace must set the hours of trading to be observed by marketplace participants.

#### 6.6.1 Trading Fees

(1) In this section, "exchange-traded fund" means a mutual fund,

- (a) the units of which are listed securities or quoted securities, and
- (b) that is in continuous distribution in accordance with applicable securities legislation.

(2) A marketplace that is subject to section 7.1 of NI 21-101 must not charge a fee for executing an order that was entered to execute against a displayed order on that marketplace greater than

(a) \$0.0030 per security traded for an equity security, or per unit traded for an exchange-traded fund, if the execution price of each security or unit traded is greater than or equal to \$1.00; or

(b) \$0.0004 per security traded for an equity security, or per unit traded for an exchange-traded fund, if the execution price of each security or unit traded is less than \$1.00.

**6.7** Anti-Avoidance - A person or company must not send an order to an exchange, quotation and trade reporting system or alternative trading system that does not carry on business in Canada in order to avoid executing against better-priced <u>protected</u> orders on a marketplace.

**6.8 Application of this Part -** In Québec, this Part, except for paragraph 6.3(1)(c), does not apply to standardized derivatives.

#### PART 7 MONITORING AND ENFORCEMENT OF REQUIREMENTS SET BY A RECOGNIZED EXCHANGE AND A RECOGNIZED QUOTATION AND TRADE REPORTING SYSTEM

#### 7.1 Requirements for a Recognized Exchange

(1) A recognized exchange must set requirements governing the conduct of its members, including requirements that the members will conduct trading activities in compliance with this Instrument.

(2) A recognized exchange must monitor the conduct of its members and enforce the requirements set under subsection (1), either

- (a) directly, or
- (b) indirectly through a regulation services provider.

(3) If a recognized exchange has entered into a written agreement under section 7.2, the recognized exchange must adopt requirements, as determined necessary by the regulation services provider, that govern the recognized exchange and the conduct of the exchange's members, and that enable the regulation services provider to effectively monitor trading on the exchange and across marketplaces.

**7.2** Agreement between a Recognized Exchange and a Regulation Services Provider - A recognized exchange that monitors the conduct of its members indirectly through a regulation services provider must enter into a written agreement with the regulation services provider which provides that the regulation services provider will:

- (a) monitor the conduct of the members of the recognized exchange,
- (b) monitor the compliance of the recognized exchange with the requirements set under subsection 7.1(3), and
- (c) enforce the requirements set under subsection 7.1(1).

**7.2.1 Obligations of a Recognized Exchange to a Regulation Services Provider** – A recognized exchange that has entered into a written agreement with a regulation services provider must

- (a) transmit to the regulation services provider the information required under Part 11 of NI 21-101 and any information reasonably required by the regulation services provider in the form and manner requested by the regulation services provider to effectively monitor:
  - (i) the conduct of and trading by marketplace participants on and across marketplaces, including the compliance of marketplace participants with the requirements set under subsection 7.1(1), and
  - (ii) the conduct of the recognized exchange, including the compliance of the recognized exchange with the requirements set under subsection 7.1(3); and
- (b) comply with all orders or directions made by the regulation services provider.

#### 7.3 Requirements for a Recognized Quotation and Trade Reporting System

(1) A recognized quotation and trade reporting system must set requirements governing the conduct of its users, including requirements that the users will conduct trading activities in compliance with this Instrument.

(2) A recognized quotation and trade reporting system must monitor the conduct of its users and enforce the requirements set under subsection (1) either

- (a) directly; or
- (b) indirectly through a regulation services provider.

(3) If a recognized quotation and trade reporting system has entered into a written agreement under section 7.4, the recognized quotation and trade reporting system must adopt requirements, as determined necessary by the regulation services provider, that govern the recognized quotation and trade reporting system's users, and that enable the regulation services provider to effectively monitor trading on the recognized quotation and trade reporting system and across marketplaces.

7.4 Agreement between a Recognized Quotation and Trade Reporting System and a Regulation Services Provider - A recognized quotation and trade reporting system that monitors the conduct of its users indirectly through a regulation services provider must enter into a written agreement with the regulation services provider which provides that the regulation services provider will

- (a) monitor the conduct of the users of the recognized quotation and trade reporting system,
- (b) monitor the compliance of the recognized quotation and trade reporting system with the requirements set under subsection 7.3(3), and
- (c) enforce the requirements set under subsection 7.3(1).

**7.4.1** Obligations of a Quotation and Trade Reporting System to a Regulation Services Provider– A recognized quotation and trade reporting system that has entered into a written agreement with a regulation services provider must

- (a) transmit to the regulation services provider the information required under Part 11 of NI 21-101 and any information reasonably required by the regulation services provider in the form and manner requested by the regulation services provider to effectively monitor:
  - (i) the conduct of and trading by marketplace participants on and across marketplaces, including the compliance of marketplace participants with the requirements set under subsection 7.3(1), and
  - (ii) the conduct of the recognized quotation and trade reporting system, including the compliance of the recognized quotation and trade reporting system with the requirements set under subsection 7.3(3); and
- (b) comply with all orders or directions made by the regulation services provider.

**7.5 Co-ordination of Monitoring and Enforcement** – A regulation services provider, recognized exchange, or recognized quotation and trade reporting system must enter into a written agreement with all other regulation services providers, recognized exchanges, and recognized quotation and trade reporting systems to coordinate monitoring and enforcement of the requirements set under Parts 7 and 8.

#### PART 8 MONITORING AND ENFORCEMENT REQUIREMENTS FOR AN ATS

**8.1 Pre-condition to Trading on an ATS** – An ATS must not execute a subscriber's order to buy or sell securities unless the ATS has executed and is subject to the written agreements required by sections 8.3 and 8.4.

#### 8.2 Requirements Set by a Regulation Services Provider for an ATS

(1) A regulation services provider must set requirements governing an ATS and its subscribers, including requirements that the ATS and its subscribers will conduct trading activities in compliance with this Instrument.

(2) A regulation services provider must monitor the conduct of an ATS and its subscribers and must enforce the requirements set under subsection (1).

**8.3** Agreement between an ATS and a Regulation Services Provider - An ATS and a regulation services provider must enter into a written agreement that provides

- (a) that the ATS will conduct its trading activities in compliance with the requirements set under subsection 8.2(1);
- (b) that the regulation services provider will monitor the conduct of the ATS and its subscribers;
- (c) that the regulation services provider will enforce the requirements set under subsection 8.2(1);
- (d) that the ATS will transmit to the regulation services provider the information required by Part 11 of NI 21-101 and any other information reasonably required to effectively monitor:
  - (i) the conduct of and trading by marketplace participants on and across marketplaces, and
  - (ii) the conduct of the ATS; and
- (e) that the ATS will comply with all orders or directions made by the regulation services provider.

**8.4** Agreement between an ATS and its Subscriber - An ATS and its subscriber must enter into a written agreement that provides

- (a) that the subscriber will conduct its trading activities in compliance with the requirements set under subsection 8.2(1);
- (b) that the subscriber acknowledges that the regulation services provider will monitor the conduct of the subscriber and enforce the requirements set under subsection 8.2(1);
- (c) that the subscriber will comply with all orders or directions made by the regulation services provider in its capacity as a regulation services provider, including orders excluding the subscriber from trading on any marketplace.
- 8.5 [Repealed]

#### PART 9 MONITORING AND ENFORCEMENT REQUIREMENTS FOR AN INTER-DEALER BOND BROKER

#### 9.1 Requirements Set by a Regulation Services Provider for an Inter-Dealer Bond Broker

(1) A regulation services provider must set requirements governing an inter-dealer bond broker, including requirements that the inter-dealer bond broker will conduct trading activities in compliance with this Instrument.

(2) A regulation services provider must monitor the conduct of an inter-dealer bond broker and must enforce the requirements set under subsection (1).

- **9.2** Agreement between an Inter-Dealer Bond Broker and a Regulation Services Provider An inter-dealer bond broker and a regulation services provider must enter into a written agreement that provides
- (a) that the inter-dealer bond broker will conduct its trading activities in compliance with the requirements set under subsection 9.1(1);
- (b) that the regulation services provider will monitor the conduct of the inter-dealer bond broker;
- (c) that the regulation services provider will enforce the requirements set under subsection 9.1(1); and
- (d) that the inter-dealer bond broker will comply with all orders or directions made by the regulation services provider.

#### 9.3 Exemption for an Inter-Dealer Bond Broker

(1) Sections 9.1 and 9.2 do not apply to an inter-dealer bond broker, if the inter-dealer bond broker complies with the requirements of IIROC Rule 2800 Code of Conduct for Corporation Dealer Member Firms Trading in Wholesale Domestic Debt Markets, as amended. (2) [Repealed]

#### PART 10 MONITORING AND ENFORCEMENT REQUIREMENTS FOR A DEALER EXECUTING TRADES OF UNLISTED DEBT SECURITIES OUTSIDE OF A MARKETPLACE

# 10.1 Requirements Set by a Regulation Services Provider for a Dealer Executing Trades of Unlisted Debt Securities Outside of a Marketplace

- (1) A regulation services provider must set requirements governing a dealer executing trades of unlisted debt securities outside of a marketplace, including requirements that the dealer will conduct trading activities in compliance with this Instrument.
- (2) A regulation services provider must monitor the conduct of a dealer executing trades of unlisted debt securities outside of a marketplace and must enforce the requirements set under subsection (1).

**10.2** Agreement between a Dealer Executing Trades of Unlisted Debt Securities Outside of a Marketplace and a Regulation Services Provider - A dealer executing trades of unlisted debt securities outside of a marketplace must enter into a written agreement with a regulation services provider that provides

- (a) that the dealer will conduct its trading activities in compliance with the requirements set under subsection 10.1(1);
- (b) that the regulation services provider will monitor the conduct of the dealer;
- (c) that the regulation services provider will enforce the requirements set under subsection 10.1(1); and
- (d) that the dealer will comply with all orders or directions made by the regulation services provider.

#### 10.3 [Repealed]

#### PART 11 AUDIT TRAIL REQUIREMENTS

#### 11.1 Application of this Part

(1) This Part does not apply to a dealer that is carrying on business as an ATS in compliance with section 6.1 of NI 21-101.

(2) A dealer or inter-dealer bond broker is exempt from the requirements in section 11.2 if the dealer or interdealer bond broker complies with similar requirements, for any securities specified, established by a regulation services provider and approved by the applicable securities regulatory authority.

#### 11.2 Audit Trail Requirements for Dealers and Inter-Dealer Bond Brokers

(1) **Recording Requirements for Receipt or Origination of an Order -** Immediately following the receipt or origination of an order for equity, fixed income and other securities identified by a regulation services provider, a dealer and inter-dealer bond broker must record in electronic form specific information relating to that order including,

- (a) the order identifier;
- (b) the dealer or inter-dealer bond broker identifier;
- (c) the type, issuer, class, series and symbol of the security;
- (d) the face amount or unit price of the order, if applicable;
- (e) the number of securities to which the order applies;
- (f) the strike date and strike price, if applicable;
- (g) whether the order is a buy or sell order;

- (h) whether the order is a short sale order, if applicable;
- (i) whether the order is a market order, limit order or other type of order, and if the order is not a market order, the price at which the order is to trade;
- (j) the date and time the order is first originated or received by the dealer or inter-dealer bond broker;
- (k) whether the account is a retail, wholesale, employee, proprietary or any other type of account;
- (I) the client account number or client identifier;
- (m) the date and time that the order expires;
- (n) whether the order is an intentional cross;
- (o) whether the order is a jitney and if so, the underlying broker identifier;
- (p) any client instructions or consents respecting the handling or trading of the order, if applicable;
- (q) the currency of the order;
- (r) an insider marker;
- (s) any other markers required by a regulation services provider;
- (t) each unique client identifier assigned to a client accessing the marketplace using direct electronic access; and
- (u) whether the order is a directed-action order.

(2) **Recording Requirements for Transmission of an Order -** Immediately following the transmission of an order for securities to a dealer, inter-dealer bond broker or a marketplace, a dealer or inter-dealer bond broker transmitting the order must add to the record of the order maintained in accordance with this section specific information relating to that order including,

- (a) the dealer or inter-dealer bond broker identifier assigned to the dealer or inter-dealer bond broker transmitting the order and the identifier assigned to the dealer, inter-dealer bond broker or marketplace to which the order is transmitted; and
- (b) the date and time the order is transmitted.

(3) **Recording Requirements for Variation, Correction or Cancellation of an Order -** Immediately following the variation, correction or cancellation of an order for securities, a dealer or inter-dealer bond broker must add to the record of the order maintained in accordance with this section specific information relating to that order including,

- (a) the date and time the variation, correction or cancellation was originated or received;
- (b) whether the order was varied, corrected or cancelled on the instructions of the client, the dealer or the interdealer bond broker;
- (c) in the case of variation or correction, any of the information required by subsection (1) which has been changed; and
- (d) the date and time the variation, correction or cancellation of the order is entered.

(4) **Recording Requirements for Execution of an Order** – Immediately following the execution of an order for securities, the dealer or inter-dealer bond broker must add to the record maintained in accordance with this section specific information relating to that order including,

(a) the identifier of the marketplace where the order was executed or the identifier of the dealer or inter-dealer bond broker executing the order if the order was not executed on a marketplace;

- (b) the date and time of the execution of the order;
- (c) whether the order was fully or partially executed;
- (d) the number of securities bought or sold;
- (e) whether the transaction was a cross;
- (f) whether the dealer has executed the order as principal;
- (g) the commission charged and all other transaction fees; and
- (h) the price at which the order was executed, including mark-up or mark-down.
- (5) [Repealed]

#### (6) [Repealed]

(7) **Record Preservation Requirements -** A dealer and an inter-dealer bond broker must keep all records in electronic form for a period of not less than seven years from the creation of the record referred to in this section, and for the first two years in a readily accessible location.

- 11.3 Transmission in Electronic Form A dealer and inter-dealer bond broker must transmit
- (a) to a regulation services provider the information required by the regulation services provider, within ten business days, in electronic form; and
- (b) to the securities regulatory authority the information required by the securities regulatory authority under securities legislation, within ten business days, in electronic form.

#### PART 12 EXEMPTION

#### 12.1 Exemption

- (1) The regulator or the securities regulatory authority may grant an exemption from this Instrument, in whole or in part, subject to such conditions or restrictions as may be imposed in the exemption.
- (2) Despite subsection (1), in Ontario, only the regulator may grant such an exemption.

#### PART 13 EFFECTIVE DATE

**13.1** Effective Date - This Instrument comes into force on December 1, 2001.

### **COMPANION POLICY 23-101CP**

### TRADING RULES

### Table of Contents

	PART	TITLE
	PART 1 1.1 1.2	INTRODUCTION Introduction Just and Equitable Principles of Trade
	PART 1.1 1.1.1 1.1.2 <u>1.1.2.1</u> 1.1.3 1.1.4 1.1.5 1.1.6 1.1.7	DEFINITIONS Definition of best execution Definition of automated <u>trading</u> functionality <u>Application to marketplaces implementing intentional order processing delays</u> Definition of protected order Definition of calculated-price order Definition of directed-action order Definition of non-standard order Definition of trade-through
	PART 2 2.1	APPLICATION OF THE INSTRUMENT Application of the Instrument
	PART 3 3.1	MANIPULATION AND FRAUD Manipulation and Fraud
	PART 4 4.1	BEST EXECUTION Best Execution
	PART 5 5.1	REGULATORY HALTS Regulatory Halts
	PART 6 6.1 6.2 6.3 6.4 6.5	ORDER PROTECTION Marketplace Requirements for Order Protection Marketplace Participant Requirements for Order Protection List of Trade-throughs Locked and Crossed Markets Anti-Avoidance Provision
	PART 7 7.1	MONITORING AND ENFORCEMENT Monitoring and Enforcement Requirements Set By a Recognized Exchange or Recognized Quotation and Trade Reporting System
	7.2 7.3 7.4 7.5	Monitoring and Enforcement Requirements for an ATS Monitoring and Enforcement Requirements for an Inter-Dealer Bond Broker Monitoring and Enforcement Requirements for a Dealer Executing Trades of Unlisted Debt Securities Outside of a Marketplace Agreement between a Marketplace and a Regulation Services Provider
	7.6 PART 8 8.1 8.2 8.3	Coordination of Monitoring and Enforcement AUDIT TRAIL REQUIREMENTS Audit Trail Requirements Transmission of Information to a Regulation Services Provider Electronic Form

#### COMPANION POLICY 23-101CP

#### TRADING RULES

#### PART 1 INTRODUCTION

**1.1** Introduction - The purpose of this Companion Policy is to state the views of the Canadian securities regulatory authorities on various matters related to National Instrument 23-101 Trading Rules (the "Instrument"), including

- (a) a discussion of the general approach taken by the Canadian securities regulatory authorities in, and the general regulatory purpose for, the Instrument; and
- (b) the interpretation of various terms and provisions in the Instrument.

**1.2** Just and Equitable Principles of Trade - While the Instrument deals with specific trading practices, as a general matter, the Canadian securities regulatory authorities expect marketplace participants to transact business openly and fairly, and in accordance with just and equitable principles of trade.

#### PART 1.1 DEFINITIONS

**1.1.1 Definition of best execution -** (1) In the Instrument, best execution is defined as the "most advantageous execution terms reasonably available under the circumstances". In seeking best execution, a dealer or adviser may consider a number of elements, including:

- a. price;
- b. speed of execution;
- c. certainty of execution; and
- d. the overall cost of the transaction.

These four broad elements encompass more specific considerations, such as order size, reliability of quotes, liquidity, market impact (i.e. the price movement that occurs when executing an order) and opportunity cost (i.e. the missed opportunity to obtain a better price when an order is not completed at the most advantageous time). The overall cost of the transaction is meant to include, where appropriate, all costs associated with accessing an order and/or executing a trade that are passed on to a client, including fees arising from trading on a particular marketplace, jitney fees (i.e. any fees charged by one dealer to another for providing trading access) and settlement costs. The commission fees charged by a dealer would also be a cost of the transaction.

(2) The elements to be considered in determining "the most advantageous execution terms reasonably available" (i.e. best execution) and the weight given to each will vary depending on the instructions and needs of the client, the particular security, the prevailing market conditions and whether the dealer or adviser is responsible for best execution under the circumstances. Please see a detailed discussion below in Part 4.

**1.1.2** Definition of automated <u>trading functionality</u> - Section 1.1 of the Instrument includes a definition of "automated <u>trading functionality</u>" which is the ability to:

- (1) act on an incoming order;
- (2) respond to the sender of an order; and
- (3) update the order by disseminating information to an information processor or information vendor.

Automated <u>trading</u> functionality allows for an incoming order to execute immediately and automatically up to the displayed size and for any unexecuted portion of such incoming order to be cancelled immediately and automatically without being booked or routed elsewhere. Automated <u>trading</u> functionality involves no human discretion in determining the action taken with respect to an order after the time the order is received. A marketplace with this functionality should have appropriate systems and policies and procedures relating to the handling of immediate-or-cancel orders.

DM#1819075

#### 1.1.2.1 Application to marketplaces implementing intentional order processing delays

(1) Paragraph (b) of the definition of "automated trading functionality" refers to the ability of a marketplace to "immediately and automatically execute an order marked as immediate-or-cancel against the displayed volume".

With respect to the application of sections 6.1 and 6.4, Canadian securities regulatory authorities are of the view that where a marketplace has introduced functionality that imposes an intentional order processing delay that is not applied in the same way to all orders, that marketplace does not provide the ability for an immediate execution against the displayed volume and therefore, does not offer "automated trading functionality". As a result, an order on that marketplace would not be a "protected order" as defined in the Instrument.

Delays in the execution of an order on a particular marketplace might result from operational or technological decisions by a marketplace. The determination of whether the marketplace with a delay offers the ability to immediately execute an order would also be based on, among other factors, how the operational model of the marketplace itself is applied, and the impact of the model or delay as it relates to fair and orderly trading. Although these delays generally would be considered intentional, they could still result in "immediate" executions on that marketplace, despite the fact that executions could be achieved faster on marketplaces that make different decisions.

If a marketplace operates more than one market or facility and it implements an intentional delay in order processing on one or more of them, only the market or facility with an intentional processing delay is considered not to provide automated trading functionality.

(2) For greater certainty, an order processing delay that is imposed solely to comply with securities legislation is not considered an intentional delay.

1.1.3 Definition of protected order - (1) A "protected order" is defined to be a "protected bid or protected offer". A "protected bid" or "protected offer" is an order to buy or sell an exchange-traded security, other than an option, that is displayed on a marketplace that provides automated trading\_functionality and about which information is provided to an information processor or an information vendor, as applicable, pursuant to Part 7 of NI 21-101. In addition, a "protected bid" or "protected offer" is a bid or offer displayed on a marketplace that meets or exceeds the market share threshold as set by the regulator, or in Quebec, the securities regulatory authority, or on a recognized exchange that does not meet the market share threshold and the bid or offer displayed is for a security listed by and traded on the recognized exchange.

(2) The regulator, or in Quebec, the securities regulatory authority, will apply the threshold on an established periodic basis to assess which marketplaces, including which markets or facilities of a marketplace, meet or exceed the market share threshold for the purposes of the definitions of "protected bid" and "protected offer". The market share threshold will be applied at the market or facility level where the marketplace is comprised of more than one visible continuous auction order book, and will not be calculated in aggregate across those different markets or facilities. A list of those that meet or exceed the market share threshold will be published on the websites of the Canadian securities regulatory authorities and the regulation services provider, so that marketplace participants can easily identify the marketplaces on which displayed orders will be considered to be protected orders in accordance with subparagraph (a)(i) of the definitions of "protected bid" and "protected offer". An updated list will be published after each periodic assessment of which marketplaces meet or exceed the market share threshold, and participants will be given an appropriate amount of time before the effective date of the published list to make any changes to operational processes that might be needed.

(3) In accordance with subsection (a)(ii) of the definitions of "protected bid" and "protected offer", a protected order is also an order displayed on a marketplace that has not met the market share threshold where that marketplace is a recognized exchange, and the order being displayed is for a security listed by and traded on the exchange. The published list will also identify any such recognized exchanges.

(4) The market share threshold criteria, including the specifics regarding the time periods covered by the calculation and the effective date and duration of the published lists, will also be made public. The application of these criteria will be monitored and reviewed, and modifications will be made if and where appropriate or necessary. Advance public notice will be made regarding any changes to the market share threshold criteria.

(2)(5) The term "displayed on a marketplace" refers to the information about total disclosed volume on a marketplace. Volumes that are not disclosed or that are "reserve" or hidden volumes are not considered to be

"displayed on a marketplace". The order must be provided in a way that enables other marketplaces and marketplace participants to readily access the information and integrate it into their systems or order routers.

(3)(6) Subsection 5.1(3) of 21-101CP does not consider orders that are not immediately executable or that have special terms as "orders" that are required to be provided to an information processor or information vendor under Part 7 of NI 21-101. As a result, these orders are not considered to be "protected orders" under the definition in the Instrument and do not receive order protection. However, those executing against these types of orders are required to execute against all better-priced protected orders first. In addition, when entering a "special terms order" on a marketplace, if it can be executed against existing protected orders despite the special term, then the order protection obligation applies.

**1.1.4 Definition of calculated-price order -** The definition of "**calculated-price order**" refers to any order where the price is not known at the time of order entry and is not based, directly or indirectly, on the quoted price of an exchange-traded security at the time the commitment to executing the order was made. This includes the following orders:

- (a) a call market order where the price of a trade is calculated by the trading system of a marketplace at a time designated by the marketplace;
- (b) an opening order where each marketplace may establish its own formula for the determination of opening prices;
- (c) a closing order where execution occurs at the closing price on a particular marketplace, but at the time of order entry, the price is not known;
- (d) a volume-weighted average price order where the price of a trade is determined by a formula that measures average price on one or more marketplaces; and
- (e) a basis order where the price is based on prices achieved in one or more derivative transactions on a marketplace. To qualify as a basis order, this order must be approved by a regulation services provider or an exchange or quotation and trade reporting system that oversees the conduct of its members or users respectively.

**1.1.5 Definition of directed-action order -** (1) An order marked as a directed-action order informs the receiving marketplace that the marketplace can act immediately to carry out the action specified by either the marketplace or marketplace participant who has sent the order and that the order protection obligation is being met by the sender. Such an order may be marked "DAO" by a marketplace or a marketplace participant. Senders can specify actions by adding markers that instruct a marketplace to:

- (a) execute the order and cancel the remainder using an immediate-or-cancel marker,
- (b) execute the order and book the remainder,
- (c) book the order as a passive order awaiting execution, and
- (d) avoid interaction with hidden liquidity using a bypass marker, as defined in IIROC's Universal Market Integrity Rules.

The definition allows for the simultaneous routing of more than one directed-action order in order to execute against any better-priced protected orders. In addition, marketplaces or marketplace participants may send a single directed-action order to execute against the best protected bid or best protected offer. When it receives a directed-action order, a marketplace can carry out the sender's instructions without checking for better-priced <u>protected</u> orders displayed by the other marketplaces and implementing the marketplace's own policies and procedures to reasonably prevent trade-throughs.

(2) Regardless of whether the entry of a directed-action order is accompanied by the bypass marker, the sender must take out all better-priced visible <u>protected</u> orders before executing at an inferior price. For example, if a marketplace or marketplace participant combines a directed-action order with a bypass marker to avoid executing against hidden liquidity, the order has order protection obligations regarding the visible <u>protected</u>

liquidity. If a directed-action order interacts with hidden liquidity, the requirement to take out all better-priced visible <u>protected</u> orders before executing at <u>an inferior pricea price that is inferior to the best protected bid or best</u> <u>protected offer</u> remains.

**1.1.6 Definition of non-standard order -** The definition of "**non-standard order**" refers to an order for the purchase or sale of a security that is subject to terms or conditions relating to settlement that have not been set by the marketplace on which the security is listed or quoted. A marketplace participant, however, may not add a special settlement term or condition to an order solely for the purpose that the order becomes a non-standard order under the definition.

**1.1.7 Definition of trade-through** -- The definition of 'trade-through' applies only to a trade executed at a price that is inferior to the best protected bid or best protected offer. It is a trade-through regardless of whether the trade occurs on a marketplace that displays protected orders, or one that does not display protected orders. For example, a trade-through would occur if executing against an order that is displayed on an ATS that does not meet the market share threshold and at a price that is inferior to the best-priced protected order. However, a trade-through would not occur if executing against a best-priced protected order despite there being a better-priced order displayed on an ATS that does not meet the market share threshold.

#### PART 2 APPLICATION OF THE INSTRUMENT

**2.1 Application of the Instrument -** Section 2.1 of the Instrument provides an exemption from subsection 3.1(1) and Parts 4 and 5 of the Instrument if a person or company complies with similar requirements established by a recognized exchange that monitors and enforces the requirements set under subsection 7.1(1) of the Instrument directly, a recognized quotation and trade reporting system that monitors and enforces requirements set under subsection 7.3(1) of the Instrument directly or a regulation services provider. The requirements are filed by the recognized exchange, recognized quotation and trade reporting system or regulation services provider and approved by a securities regulatory authority. If a person or company is not in compliance with the requirements of the recognized exchange, recognized quotation and trade reporting system or the regulation services provider, then the exemption does not apply and that person or company is subject to subsection 3.1(1) and Parts 4 and 5 of the Instrument. The exemption from subsection 3.1(1) does not apply in Alberta, British Columbia, Ontario, Québec and Saskatchewan and the relevant provisions of securities legislation apply.

#### PART 3 MANIPULATION AND FRAUD

#### 3.1 Manipulation and Fraud

(1) Subsection 3.1(1) of the Instrument prohibits the practices of manipulation and deceptive trading, as these may create misleading price and trade activity, which are detrimental to investors and the integrity of the market.

(2) Subsection 3.1(2) of the Instrument provides that despite subsection 3.1(1) of the Instrument, the provisions of the *Securities* Act (Alberta), the *Securities* Act (British Columbia), the *Securities* Act (Ontario), the *Securities* Act (Québec) and *The Securities* Act, 1988 (Saskatchewan), respectively, relating to manipulation and fraud apply in Alberta, British Columbia, Ontario, Québec and Saskatchewan. The jurisdictions listed have provisions in their legislation that deal with manipulation and fraud.

(3) For the purposes of subsection 3.1(1) of the Instrument, and without limiting the generality of those provisions, the Canadian securities regulatory authorities, depending on the circumstances, would normally consider the following to result in, contribute to or create a misleading appearance of trading activity in, or an artificial price for, a security:

- (a) Executing transactions in a security if the transactions do not involve a change in beneficial or economic ownership. This includes activities such as wash-trading.
- (b) Effecting transactions that have the effect of artificially raising, lowering or maintaining the price of the security. For example, making purchases of or offers to purchase securities at successively higher prices or making sales of or offers to sell a security at successively lower prices or entering an order or orders for the purchase or sale of a security to:
  - (i) establish a predetermined price or quotation,

- (ii) effect a high or low closing price or closing quotation, or
- (iii) maintain the trading price, ask price or bid price within a predetermined range.
- (c) Entering orders that could reasonably be expected to create an artificial appearance of investor participation in the market. For example, entering an order for the purchase or sale of a security with the knowledge that an order of substantially the same size, at substantially the same time, at substantially the same price for the sale or purchase, respectively, of that security has been or will be entered by or for the same or different persons.
- (d) Executing prearranged transactions that have the effect of creating a misleading appearance of active public trading or that have the effect of improperly excluding other marketplace participants from the transaction.
- (e) Effecting transactions if the purpose of the transactions is to defer payment for the securities traded.
- (f) Entering orders to purchase or sell securities without the ability and the intention to
  - (i) make the payment necessary to properly settle the transaction, in the case of a purchase; or
  - (ii) deliver the securities necessary to properly settle the transaction, in the case of a sale.

This includes activities known as free-riding, kiting or debit kiting, in which a person or company avoids having to make payment or deliver securities to settle a trade.

- (g) Engaging in any transaction, practice or scheme that unduly interferes with the normal forces of demand for or supply of a security or that artificially restricts or reduces the public float of a security in a way that could reasonably be expected to result in an artificial price for the security.
- (h) Engaging in manipulative trading activity designed to increase the value of a derivative position.
- (i) Entering a series of orders for a security that are not intended to be executed.

(4) The Canadian securities regulatory authorities do not consider market stabilization activities carried out in connection with a distribution to be activities in breach of subsection 3.1(1) of the Instrument, if the market stabilization activities are carried out in compliance with the rules of the marketplace on which the securities trade or with provisions of securities legislation that permit market stabilization by a person or company in connection with a distribution.

(5) Section 3.1 of the Instrument applies to transactions both on and off a marketplace. In determining whether a transaction results in, contributes to or creates a misleading appearance of trading activity in, or an artificial price for a security, it may be relevant whether the transaction takes place on or off a marketplace. For example, a transfer of securities to a holding company for *bona fide* purposes that takes place off a marketplace would not normally violate section 3.1 even though it is a transfer with no change in beneficial ownership.

(6) The Canadian securities regulatory authorities are of the view that section 3.1 of the Instrument does not create a private right of action.

(7) In the view of the Canadian securities regulatory authorities, section 3.1 includes attempting to create a misleading appearance of trading activity in or an artificial price for, a security or attempting to perpetrate a fraud.

### PART 4 BEST EXECUTION

#### 4.1 Best Execution

(1) The best execution obligation in Part 4 of the Instrument does not apply to an ATS that is registered as a dealer provided that it is carrying on business as a marketplace and is not handling any client orders other than accepting them to allow them to execute on the system. However, the best execution obligation does otherwise apply to an ATS acting as an agent for a client.

(2) Section 4.2 of the Instrument requires a dealer or adviser to make reasonable efforts to achieve best execution (the most advantageous execution terms reasonably available under the circumstances) when acting for a client. The obligation applies to all securities.

#### DM#1819075

(3) Although wWhat constitutes "best execution" varies will vary depending on the particular circumstances, and is subject to a "reasonable efforts" test that does not require achieving best execution for each and every teorder. To meet the "reasonable efforts" test, a dealer or adviser should be able to demonstrate that it has, and has abided by, its policies and procedures that (i) require it to follow the client's instructions and the objectives set, and (ii) outline a the process it has designed toward the objective of achievinge best execution. The policies and procedures should be regularly and rigorously reviewed. The policies outlining the obligations of the dealer or adviser will be dependent on the role it is playing in an execution. For example, in making reasonable efforts to achieve best execution, the dealer should consider the client's instructions and a number of factors, including the client's investment objectives and the dealer's knowledge of markets and trading patterns. An adviser should consider a number of factors, including assessing a particular client's requirements or portfolio objectives, selecting appropriate dealers and marketplaces and monitoring the results on a regular basis. In addition, if an adviser is directly accessing a marketplace, the factors to be considered by dealers may also be applicable.

(4) Where securities listed on a Canadian exchange or quoted on a Canadian quotation and trade reporting system are inter-listed either within Canada or on a foreign exchange or quotation and trade reporting system, in making reasonable efforts to achieve best execution, the dealer should assess whether it is appropriate to consider all marketplaces upon which the security is listed or quoted and where the security is traded, both within and outside of Canada.

(5) In order to meet best execution obligations where securities trade on multiple marketplaces in Canada, a dealer should consider information from all appropriate marketplaces, and (not just marketplaces where the dealer is a participant). This does not mean that a dealer must have access to real-time data feeds from each marketplace. However, its policies and procedures for seeking best execution should include the process for taking into account order and/or trade information from allconsidering activity on appropriate marketplaces and an evaluation of whether steps should be taken to the requirement to evaluate whether taking steps to access orders is appropriate under the circumstances on a marketplace to which it does not have access. The steps to access orders may include making arrangements with another dealer who is a participant of a particular marketplace.

(6) As part of an evaluation of whether steps should be taken to access orders on a marketplace to which it does not have access, a dealer should consider how the decision to access or not access orders on that marketplace will impact its ability to achieve best execution for its clients, taking into consideration those clients' objectives and needs. This applies in relation to decisions as to whether to access marketplaces that do not provide pre-trade transparency of orders, as well as those that do display orders that are not protected orders. We expect that documented best execution policies and procedures would include the rationale for accessing or not accessing orders on particular marketplaces, and that the rationale will be reviewed for continued reasonableness at least annually, and more frequently if needed because of changes to the trading environment and market structure. This review might require an analysis of historical data relating to the order and trade activity on marketplaces to which the dealer does not have access. We expect that the factors to be considered in such an analysis would generally include the frequency at which a better price is available, size and depth of quotes, traded volumes, potential market impact, and market share (considering the types and classes of securities traded by clients, generally).

(67) For foreign exchange-traded securities, if they are traded on a marketplace in Canada, dealers should include in their best execution policies and procedures a regular assessment of whether it is appropriate to consider the marketplace as well as the foreign markets upon which the securities trade.

(78) Section 4.2 of the Instrument applies to registered advisers as well as registered dealers that carry out advisory functions but are exempt from registration as advisers.

(8<u>9</u>) Section 4.3 of the Instrument requires that a dealer or adviser make reasonable efforts to use facilities providing information regarding orders and trades. These reasonable efforts refer to the use of the information displayed by the information processor or, if there is no information processor, an information vendor.

### PART 5 REGULATORY HALTS

**5.1 Regulatory Halts** - Section 5.1 of the Instrument applies when a regulatory halt has been imposed by a regulation services provider, a recognized exchange, or a recognized quotation and trade reporting system. A regulatory halt, as referred to in section 5.1 of the Instrument, is one that is imposed to maintain a fair and orderly market, including halts related to a timely disclosure policy, or because there has been a violation of regulatory requirements. In the view of the Canadian securities regulatory authorities, an order may trade on a marketplace

despite the fact that trading of the security has been suspended because the issuer of the security has ceased to meet minimum listing or quotation requirements, or has failed to pay to the recognized exchange, or the recognized quotation and trade reporting system any fees in respect of the listing or quotation of securities of the issuer. Similarly, an order may trade on a marketplace despite the fact that trading of the security has been delayed or halted because of technical problems affecting only the trading system of the recognized exchange, or recognized quotation and trade reporting system.

### PART 6 ORDER PROTECTION

#### 6.1 Marketplace Requirements for Order Protection

(1) Subsection 6.1(1) of the instrument requires a marketplace to establish, maintain and ensure compliance with written policies and procedures that are reasonably designed to prevent trade-throughs by orders entered on that marketplace, regardless of whether the marketplace on which that order is entered displays orders that are protected orders. A marketplace may implement this requirement in various ways. For example, the policies and procedures of a marketplace may reasonably prevent trade-throughs via the design of the marketplace's trade execution algorithms (by not allowing a trade-through to occur), or by voluntarily establishing direct linkages to other marketplaces. Marketplaces are not able to avoid their obligations by establishing policies and procedures that instead require marketplace participants to take steps to reasonably prevent trade-throughs.

(2) It is the responsibility of marketplaces to regularly review and monitor the effectiveness of their policies and procedures and take prompt steps to remedy any deficiencies in reasonably preventing trade-throughs and complying with subsection 6.1(2) of the Instrument. In general, it is expected that marketplaces maintain relevant information so that the effectiveness of its policies and procedures can be adequately evaluated by regulatory authorities. Relevant information would include information that describes:

- (a) steps taken by the marketplace to evaluate its policies and procedures;
- (b) any breaches or deficiencies found; and
- (c) the steps taken to resolve the breaches or deficiencies.

(3) As part of the policies and procedures required in subsection 6.1(1) of the Instrument, a marketplace is expected to include a discussion of their automated <u>trading</u> functionality and how they will handle potential delayed responses as a result of an equipment or systems failure or malfunction experienced by any other marketplace <u>displaying protected orders</u>. In addition, marketplaces should include a discussion of how they treat a directed-action order when received and how it will be used.

(4) Order protection applies whenever two or more marketplaces with that display orders subject to the pretrade transparency requirements in Part 7 of NI 21-101 are open for trading, and the displayed orders of at least one of those marketplaces are protected orders are open for trading. Some marketplaces provide a trading session at a price established by that marketplace during its regular trading hours for marketplace participants who are required to benchmark to a certain closing price. In these circumstances, under <u>sub</u>paragraph 6.2(e)(iii) of the Instrument, a marketplace <u>that provides such sessions</u> would not be required to take steps to reasonably prevent trade-throughs of <u>protected</u> orders on another marketplace.

### 6.2 Marketplace Participant Requirements for Order Protection

(1) For a marketplace participant that wants to use a directed-action order, section 6.4 of the Instrument requires a marketplace participant to establish, maintain and ensure compliance with written policies and procedures that are reasonably designed to prevent trade-throughs of protected orders, regardless of whether the marketplace on which it is entering the directed-action order displays orders that are protected orders. In general, it is expected that a marketplace participant that uses a directed-action order would maintain relevant information so that the effectiveness of its policies and procedures can be adequately evaluated by regulatory authorities. Relevant information would include information that describes:

- (a) steps taken by the marketplace participant to evaluate its policies and procedures;
- (b) any breaches or deficiencies found; and
- (c) the steps taken to resolve the breaches or deficiencies.

The policies and procedures should also outline when it is appropriate to use a directed-action order and how it will be used as set out in paragraph 6.4(1)(a) of the Instrument.

(2) Order protection applies whenever two or more marketplaces with\_that display orders subject to the pretrade transparency requirements in Part 7 of NI 21-101 are open for trading, and the displayed orders of at least one of those marketplaces are protected orders are open for trading. Some marketplaces provide a trading session at a price established by that marketplace during its regular trading hours for marketplace participants who are required to benchmark to a certain closing price. In these circumstances, under paragraph 6.4(1)(a)(iv)(C) of the Instrument, a marketplace participant would not be required to take steps to reasonably prevent trade-throughs of <u>protected</u> orders on other marketplaces that result from an execution of the closing-price order<del>between marketplaces</del>.

**6.3** List of Trade-throughs - Section 6.2 and paragraphs 6.4(1)(a)(i) to 6.4(1)(a)(v) of the Instrument set forth a list of "permitted" trade-throughs that are primarily designed to achieve workable order protection and to facilitate certain trading strategies and order types that are useful to investors.

(a) (i) Paragraphs 6.2(a) and 6.4(1)(a)(i) of the Instrument would apply where a marketplace or marketplace participant, as applicable, has reasonably concluded that <u>a the marketplace displaying the protected order that has been traded through is experiencing a failure, malfunction or material delay of its systems, equipment or ability to disseminate marketplace data. A material delay occurs when a marketplace repeatedly fails to respond immediately after receipt of an order. This is intended to provide marketplaces and marketplace participants with flexibility when dealing with a marketplace that is experiencing systems problems (either of a temporary nature or a longer term systems issue).</u>

Under subsection 6.3(1) of the Instrument, a marketplace that is experiencing systems issues is (ii) responsible for informing all other marketplaces, its marketplace participants, any information processor, or if there is no information processor, an information vendor disseminating its information under Part 7 of NI 21-101 and regulation services providers when a failure, malfunction or material delay of its systems. equipment or ability to disseminate marketplace data occurs. This applies both the-to marketplaces that display orders that are protected orders and marketplaces that display orders that are not protected orders. However, if a marketplace that displays orders that are protected orders fails repeatedly to provide an immediate response to orders received and no notification has been issued by that marketplace that it is experiencing systems issues, the routing marketplace or a marketplace participant may, pursuant to subsections 6.3(2) and 6.3(3) of the Instrument respectively, reasonably conclude that the marketplace is having systems issues and may therefore rely on paragraph 6.2(a) or 6.4(1)(a)(i) of the Instrument respectively. This reliance must be done in accordance with policies and procedures that outline processes for dealing with potential delays in responses by a marketplace and documenting the basis of its conclusion. If, in response to the notification by the routing marketplace or a marketplace participant, the marketplace confirms that it is not actually experiencing systems issues, the routing marketplace or marketplace participant may no longer rely on paragraph 6.2(a) or paragraph 6.4(1)(a)(i) of the Instrument respectively.

- (b) Paragraph 6.2(b) of the Instrument provides an exception from the obligation on marketplaces to use their policies and procedures to reasonably prevent trade-throughs when a directed-action order is received. Specifically, a marketplace that receives a directed-action order may immediately execute or book the order (or its remaining volume) and not implement the marketplace's policies and procedures to reasonably prevent trade-throughs. However, the marketplace will need to describe its treatment of a directed-action order in its policies and procedures. Paragraphs 6.2(c) and 6.4(1)(a)(ii) of the Instrument provide an exception where a marketplace or marketplace participant simultaneously routes directed-action orders to execute against the total displayed volume of any protected order traded through. This accounts for the possibility that orders that are routed simultaneously as directed-action orders are not executed simultaneously causing one or more trade-throughs to occur because an inferior-priced order is executed first.
- (c) Paragraphs 6.2(d) and 6.4(1)(a)(ii) of the Instrument provide some relief due to moving or changing markets. Specifically, the exception allows for a trade-through to occur when immediately before executing the order that caused the trade-through, the marketplace on which the execution occurred had the best price but at the moment of execution, the market changes and another marketplace has the best price<u>d protected order</u>. The "changing markets" exception allows for the execution of an order on a marketplace, within the best bid or offer on that marketplace but outside the best<u>protected</u> bid or <u>best protected</u> offer displayed across <u>all</u> marketplaces<u>that display protected orders</u>, in certain circumstances. This could occur for example:

(i) where orders are entered on a marketplace but by the time they are executed, the best <u>protected</u> bid or <u>best protected</u> offer displayed across marketplaces changed; and

(ii) where a trade is agreed to off-marketplace and entered on a marketplace within the best <u>protected</u> bid and best <u>protected</u> offer across marketplaces, but by the time the order is executed on the marketplace (i.e. printed) the best <u>protected</u> bid or <u>best protected</u> offer as displayed across marketplaces may have changed, thus causing a trade-through.

- (d) The basis for the inclusion of calculated-price orders, non-standard orders and closing-price orders in paragraphs 6.2(e) and 6.4(1)(a)(iv) of the Instrument is that these orders have certain unique characteristics that distinguish them from other orders. The characteristics of the orders relate to price (calculated-price orders and closing-price orders) and non-standard settlement terms (non-standard orders) that are not set by an exchange or a quotation and trade reporting system.
- (e) Paragraph 6.2(f) and 6.4(1)(a)(v) of the Instrument include a transaction that occurred when there is a crossed market between protected orders in the exchange-traded security. Without this allowance, no marketplace could execute transactions in a crossed marketwhere the best protected bid and best protected offer are crossed because it would constitute a trade-through. With order protection only applying to displayed protected orders or parts of protected orders, hidden or reserve orders may remain in the book after all displayed protected orders are executed. Consequently, crossed markets between protected orders may occur. Intentionally crossing the market best protected bid or best protected offer to take advantage of paragraphs 6.2(f) and 6.4(1)(a)(v) of the Instrument would be a violation of section 6.5 of the Instrument.

#### 6.4 Locked and Crossed Markets

(1) Section 6.5 of the Instrument provides that a marketplace participant or a marketplace that routes or reprices orders must not intentionally lock or cross a market-protected order by entering a protected displayed order on any marketplace to either buy a security at a price that is the same as or higher than the best protected offer or entering a protected order to sell a security at a price that is the same as or lower than the best protected bid. The intention of section 6.5 of the Instrument is to prevent intentional locks and crosses of protected orders. This applies regardless of whether the locking or crossing order is entered on a marketplace that displays orders that are protected orders. This provision is not intended to prohibit the use of marketable limit orders. Paragraphs 6.2(f) and 6.4(1)(a)(v) of the Instrument allow for the resolution of crossed markets that occur unintentionally.

The Canadian securities regulatory authorities consider an order that is routed or repriced to be "entered" on a marketplace. The Canadian securities regulatory authorities do not consider the triggering of a previously-entered onstop order to be an "entry" or "repricing" of that order.

(2) Section 6.5 of the Instrument does not restrict the ability for a marketplace participant or a marketplace that routes or reprices orders from routing or entering a displayed order that will lock or cross with another displayed order that is not a protected order.

If the entry of a protected order locks or crosses with a displayed order on another marketplace that is not a protected order, section 6.5 of the Instrument would restrict the ability for additional orders to be entered that would lock or cross with the protected order. This should help to minimize the duration of a locked or crossed markets in these circumstances.

A displayed order that is not a protected order that becomes locked or crossed with a subsequently entered protected order does not need to be repriced or cancelled. If, however, the marketplace subsequently reprices the non-protected displayed order, as might occur with a pegged order, it will be considered to be "entered" upon repricing and subject to the restrictions against locking or crossing with a protected order.

If a marketplace participant deliberately attempts to circumvent section 6.5 of the Instrument by first entering a displayed order on a marketplace that is not a protected order, followed by the entry of a protected order on another marketplace that locks or crosses with the first displayed non-protected order it entered, the Canadian securities regulatory authorities would consider this to be a violation of section 6.5.

(23) Section 6.5 of the Instrument prohibits a marketplace participant or a marketplace that routes or reprices orders from intentionally locking or crossing a market. This would occur, for example, when a marketplace participant enters a locking or crossing order on a particular marketplace or marketplaces to avoid fees charged by a marketplace or to take advantage of rebates offered by a particular marketplace. This <u>An intentional locking or crossing of a protected order</u> could also-occur where a marketplace system is programmed to reprice orders without checking to see if the new price would lock the market<u>a protected order</u> or where the marketplace routes orders to

another marketplace that results in a locked market with a protected order. It could also occur where the intention of the marketplace participant was to lock or cross a protected order to avoid fees charged by a marketplace or to take advantage of rebates.

There are situations where a locked or crossed market of a protected order may occur unintentionally. For example:

- (a) when a marketplace participant routes multiple directed action orders that are marked immediate or cancel to a variety of marketplaces and because of latency issues, a locked or crossed market results,
- (b) the locking or crossing order was displayed at a time when the marketplace displaying the locked or crossed protected order was experiencing a failure, malfunction or material delay of its systems, equipment or ability to disseminate marketplace data,
- (eb) the locking or crossing order was displayed at a time when a protected bid was higher than a protected offer;
  - (dc) the locking or crossing order was posted after all displayed <u>protected</u> liquidity was executed and a reserve order generated a new visible <u>protected</u> bid above the displayed <u>protected</u> offer or <u>new visible protected</u> offer or <u>new visible protected</u> bid-;
  - (ed) the locking or crossing order was entered on a particular marketplace in order to comply with securities legislation requirements such as Rule 904 of Regulation S of the *Securities Act of 1933* that requires securities subject to resale restrictions in the United States to be sold in Canada on a "designated offshore securities market"
  - (fe) the locking or crossing order was displayed due to "race conditions" when competing orders, at least one of which is a protected order, are entered on marketplaces at essentially the same time with neither party having knowledge of the other order at the time of entry;
  - $(\underline{gf})$  the locking or crossing order was a result of the differences in processing times and latencies between the systems of the marketplace participant, marketplaces, information processor and information vendors<sub> $\tau_i</sub>$ :</sub>
  - (hg) the locking or crossing order was a result of marketplaces having different mechanisms to "restart" trading following a halt in trading for either regulatory or business purposes, and
  - (<u>hi</u>) the locking or crossing order was a result of the execution of an order during the opening or closing allocation process of one market, while trading is simultaneously occurring on a continuous basis on another market <u>displaying protected orders</u>,

If a marketplace participant using a directed-action order chooses to book the order, or the remainder of the order not immediately executed, then it is responsible for ensuring that the booked portion of the directed-action order does not lock or cross the marketa protected order. The Canadian securities regulatory authorities would consider a directed-action order or remainder of directed-action order that is booked and that locks or crosses the marketa protected order to be an intentional locking or crossing of the marketa protected order and a violation of section 6.5 of the Instrument.

**6.5 Anti-Avoidance Provision** - Section 6.7 of the Instrument prohibits a person or company from sending an order to an exchange, quotation and trade reporting system or alternative trading system that does not carry on business in Canada in order to avoid executing against better-priced <u>protected</u> orders on a marketplace in Canada. The intention of this section is to prevent the routing of orders to foreign marketplaces only for the purpose of avoiding the order protection regime in Canada.

### PART 7 MONITORING AND ENFORCEMENT

7.1 Monitoring and Enforcement of Requirements Set By a Recognized Exchange or Recognized Quotation and Trade Reporting System - Under section 7.1 of the Instrument, a recognized exchange will set its own requirements governing the conduct of its members. Under section 7.3 of the Instrument, a recognized quotation and trade reporting system will set its own requirements governing the conduct of its users. The recognized exchange or recognized quotation and trade reporting system can monitor and enforce these requirements either directly or indirectly through a regulation services provider. A regulation services provider is a person or company that provides regulation services and is either a recognized exchange, recognized quotation and trade reporting system or a recognized self-regulatory entity.

If a recognized exchange or recognized quotation and trade reporting system has entered into a written agreement with a regulation services provider, it is expected that the requirements adopted by the recognized exchange or recognized quotation and trade reporting system under Part 7 of the Instrument will consist of all of the rules of the regulation services provider that relate to trading. For example, if a recognized exchange or recognized quotation and trade reporting system has entered into a written agreement with IIROC, the rules adopted by the recognized exchange or recognized quotation and trade reporting system are all of IIROC's Universal Market Integrity Rules. Clock synchronization, trade markers and trading halt requirements would be examples of these adopted rules that relate to the regulation services provider's monitoring of trading on the recognized exchange or recognized quotation and trade reporting system and across marketplaces.

We are of the view that all of the rules of the regulation services provider related to trading must be adopted by a recognized exchange or recognized quotation and trade reporting system that has entered into a written agreement with the regulation services provider given the importance of these rules in the context of effectively monitoring trading on and across marketplaces. We note that the regulation services provider is required to monitor the compliance of, and enforce, the adopted rules as against the members of the recognized exchange or users of the recognized quotation and trade reporting system. The regulation services provider is also required to monitor the compliance of the recognized exchange or recognized quotation and trade reporting system with the adopted rules but it is the applicable securities regulatory authority that will enforce these rules against the recognized exchange or recognized quotation and trade reporting system.

Sections 7.2 and 7.4 of the Instrument require the recognized exchange or recognized quotation and trade reporting system that chooses to have the monitoring and enforcement performed by the regulation services provider to enter into an agreement with the regulation services provider in which the regulation services provider agrees to enforce the requirements of the recognized exchange or recognized quotation and trade reporting system adopted under subsection 7.1(1) and 7.3(1).

Specifically, sections 7.2 and 7.4 require the written agreement between a recognized exchange or recognized quotation and trade reporting system and its regulation services provider to provide that the regulation services provider will monitor and enforce the requirements set under subsection 7.1(1) or 7.3(1) and monitor the requirements adopted under subsection 7.1(3) or 7.3(3).

Paragraph 7.2.1(a)(i) mandates that a recognized exchange must transmit information reasonably required by the regulation services provider to effectively monitor the conduct of and trading by marketplace participants on and across marketplaces. The reference to monitoring trading "across marketplaces" refers to the instance where particular securities are traded on multiple marketplaces. Where particular securities are only traded on one marketplace, the reference to "across marketplaces" may not apply in all circumstances.

Paragraph 7.2.1(a)(ii) requires that a recognized exchange must transmit information reasonably required by the regulation services provider to effectively monitor the compliance of the recognized exchange with the requirements adopted under subsection 7.1(3). As well, subsection 7.2.1(b) requires a recognized exchange to comply with all orders or directions of its regulation services provider that are in connection with the conduct and trading by the recognized exchange's members on the recognized exchange and with the regulation services provider's oversight of the compliance of the recognized exchange with the requirements adopted under 7.1(3).

7.2 Monitoring and Enforcement Requirements for an ATS - Section 8.2 of the Instrument requires the regulation services provider to set requirements that govern an ATS and its subscribers. Before executing a trade for a subscriber, the ATS must enter into an agreement with a regulation services provider and an agreement with each subscriber. These agreements form the basis upon which a regulation services provider will monitor the trading activities of the ATS and its subscribers and enforce its requirements. The requirements set by a regulation services provider must include requirements that the ATS and its subscribers will conduct trading activities in compliance with the Instrument. The ATS and its subscribers are considered to be in compliance with the Instrument and are exempt from the application of most of its provisions if the ATS and the subscriber are in compliance with the requirements set by a regulation services provider.

**7.3 Monitoring and Enforcement Requirements for an Inter-Dealer Bond Broker** - Section 9.1 of the Instrument requires that a regulation services provider set requirements governing the conduct of an inter-dealer bond broker. Under section 9.2 of the Instrument, the inter-dealer bond broker must enter into an agreement with the regulation services provider providing that the regulation services provider monitor the activities of the inter-dealer bond broker and enforce the requirements set by the regulation services provider. However, section 9.3 of the Instrument provides inter-dealer bond brokers with an exemption from sections 9.1 and 9.2 of the Instrument if the inter-dealer bond broker complies with the requirements of IIROC Rule 2800 Code of Conduct for Corporation

Dealer Member Firms Trading in Wholesale Domestic Debt Markets, as amended, as if that policy was drafted to apply to the inter-dealer bond broker.

7.4 Monitoring and Enforcement Requirements for a Dealer Executing Trades of Unlisted Debt Securities Outside of a Marketplace - Section 10.1 of the Instrument requires that a regulation services provider set requirements governing the conduct of a dealer executing trades of unlisted debt securities outside of a marketplace. Under section 10.2 of the Instrument, the dealer must also enter into an agreement with the regulation services provider monitor the activities of the dealer and enforce the requirements set by the regulation services provider.

#### 7.5 Agreement between a Marketplace and a Regulation Services Provider

The purpose of subsections 7.2(c) and 7.4(c) of the Instrument is to facilitate the monitoring of trading by marketplace participants on and across multiple marketplaces by a regulation services provider. These sections of the Instrument also facilitate monitoring of the conduct of a recognized exchange and recognized quotation and trade reporting system for particular purposes. This may result in regulation services providers monitoring marketplaces that have retained them and reporting to a recognized exchange, recognized quotation and trade reporting system or securities regulatory authority if a marketplace is not meeting regulatory requirements or the terms of its own rules or policies and procedures. While the scope of this monitoring may change as the market evolves, we expect it to include, at a minimum, monitoring clock synchronization, the inclusion of specific designations, symbols and identifiers, order protection requirements and audit trail requirements.

### 7.6 Coordination of Monitoring and Enforcement

(1) Section 7.5 of the Instrument requires regulation services providers, recognized exchanges and recognized quotation and trade reporting systems to enter into a written agreement whereby they coordinate the enforcement of the requirements set under Parts 7 and 8. This coordination is required in order to achieve cross-marketplace monitoring.

(2) If a recognized exchange or recognized quotation and trade reporting system has not retained a regulation services provider, it is still required to coordinate with any regulation services provider and other exchanges or quotation and trade reporting systems that trade the same securities in order to ensure effective cross-marketplace monitoring.

(3) Currently, only IIROC is the regulation services provider for both exchange-traded securities, other than options and in Québec, other than standardized derivatives, and unlisted debt securities. If more than one regulation services provider regulates marketplaces trading a particular type of security, these regulation services providers must coordinate monitoring and enforcement of the requirements set.

### PART 8 AUDIT TRAIL REQUIREMENTS

**8.1 Audit Trail Requirements -** Section 11.2 of the Instrument imposes obligations on dealers and inter-dealer bond brokers to record in electronic form and to report certain items of information with respect to orders and trades. Information to be recorded includes any markers required by a regulation services provider (such as a significant shareholder marker). The purpose of the obligations set out in Part 11 is to enable the entity performing the monitoring and surveillance functions to construct an audit trail of order, quotation and transaction data which will enhance its surveillance and examination capabilities.

**8.2** Transmission of Information to a Regulation Services Provider - Section 11.3 of the Instrument requires that a dealer and an inter-dealer bond broker provide to the regulation services provider information required by the regulation services provider, within ten business days, in electronic form. This requirement is triggered only when the regulation services provider sets requirements to transmit information.

**8.3** Electronic Form - Subsection 11.3 of the Instrument requires any information required to be transmitted to the regulation services provider and securities regulatory authority in electronic form. Dealers and inter-dealer bond brokers are required to provide information in a form that is accessible to the securities regulatory authorities and the regulation services provider (for example, in SELECTR format).

# ANNEX D

### **List of Commenters**

### 2014 Notice

Acumen Capital Finance Partners Limited Aequitas Neo Exchange Inc. BlackRock Asset Management Canada Limited **BMO** Capital Markets Canadian Advocacy Council for Canadian CFA Institute Societies Canadian Securities Exchange Canadian Securities Traders Association, Inc. Chi-X Canada ATS Limited CIBC World Markets Inc. Investment Industry Association of Canada ITG Canada Corp. Katya Malinova and Andreas Park, Department of Economics, University of Toronto Leede Financial Liquidnet Canada National Bank Financial Inc. Omega Securities Inc. **RBC** Capital Markets **RBC** Dominion Securities Inc. **RBC** Global Asset Management Scotia Capital Inc. Stikeman Elliot LLP TD Asset Management **TD** Securities TMX Group Limited Tradebot Systems Inc. True North Vantage W.D. Latimer Co. Limited

## 2015 Notice

Aequitas NEO Exchange Inc. Canadian Securities Traders Association, Inc. Chi-X Canada ATS Limited CIBC World Markets Inc. ITG Canada Corp. KOR Group LLC National Bank Financial Inc. PI Financial Corp. RBC Dominion Securities Inc. RBC Global Asset Management Inc. Scotia Capital Inc. The Canadian Advocacy Council for Canadian CFA Institute Societies The Investment Industry Association of Canada TMX Group Limited

# ANNEX E

Торіс	Summary of Comments	CSA Response
Market Share Threshold	Most commenters expressed views on the threshold proposal, and the responses were mixed.	
Threshold Level	A number of commenters supported the threshold as proposed, at 5%. These commenters indicated that, as proposed, the threshold strikes a reasonable balance and allows dealers flexibility in making marketplace connectivity decisions, while maintaining most of the benefits of OPR. However, some of those who were generally supportive of the market share threshold expressed concern with the proposed threshold level. Specific concerns were related to the ability of unprotected markets to meet and/or maintain a 5% threshold, and potential negative impacts on competition. Some suggested revising the threshold to a lower level. A number of commenters were not supportive of either the introduction of a market share threshold or the proposed level. They raised concerns related to the complexity that would be introduced by the threshold approach as well as the potential impact on competition. In addition, one commenter suggested that the threshold, as proposed, may exacerbate the problems that the CSA are looking to solve.	We acknowledge that the comments received in relation to the market share threshold were mixed, both with respect to the introduction of a threshold and the proposed level. However, we continue to be of the view that the threshold approach is the most appropriate method of balancing the benefits of OPR with some of the inefficiencies and costs associated with its implementation. In the 2014 Notice we proposed a 5% market share threshold. Based on the comments received and further discussions, we are adjusting the threshold from 5% to 2.5%. We are of the view that a lower threshold addresses some of the concerns raised regarding the potential impact on competition. With respect to the complexity introduced into the system by the threshold, we note that evolving marketplace trading models related to order processing delays have already driven regulatory decisions that introduced many of the same complexities highlighted by

# Summary of Comments and CSA Responses to 2014 Notice

		commenters
		commenters.
Effect of Threshold on Competition	With respect to the impact of the proposed threshold approach, comments were mixed. However, as noted above many commenters expressed concern that the proposed 5% threshold could have a negative impact on competition and could represent a barrier to entry for new competitors.	As noted above, we are reducing the threshold from 5% to 2.5%. We believe the threshold is important to balance some of the inefficiencies and costs of OPR, but acknowledge the concerns that in the context of the current market environment in Canada, 5% may be a barrier to competition.
Criteria for Calculating the Threshold	With respect to the metrics used in relation to the threshold, comments were mixed. Comments were generally supportive of the trading activity to be included in the calculation, but there was less consensus on the equal weighting of volume and value.	We continue to believe that equally weighting volume and value traded is appropriate in order to provide a more balanced outcome and account for marketplaces that trade primarily low-value securities.
Market share Threshold - Treatment of Listing Exchanges	Some commenters supported the proposed approach with respect to the treatment of listing exchanges that do not otherwise meet the market share threshold. Others however, expressed concern about the potential incentive for exchanges to list securities for the purposes of gaining order protection.	We continue to be of the view that it is important to provide protection to a recognized exchange that does not meet the market share threshold, but only with respect to its own listings. As indicated in the 2014 Notice, we do not believe that it is appropriate that a listing exchange be disadvantaged with respect to its own listings, to the advantage of a marketplace that might trade very little volume in those listed securities, but that otherwise meets the threshold. Further, we are supportive of continued protection for exchanges that contribute to the capital raising process.

		We acknowledge the concerns raised by commenters related to an exchange seeking cross- listings simply for purposes of gaining OPR protection. We intend to monitor trends in this regard going forward.
Time Frame for Market Share Calculation	Comments were mixed, with some supportive of the annual calculation proposed in the 2014 Notice and the three month implementation window. Others however, were concerned that the annual calculation was too long, and that marketplaces meeting the threshold during the annual period would be required to wait too long for protected status. Some suggested either a more frequent review period or a decrease in the threshold level.	As noted above, we have reduced the threshold from 5% to 2.5%, and believe this will address some of the concerns raised. However, we remain supportive of an annual measurement in order to decrease the costs on industry that might arise from more frequent calculations and potential changes to the list of protected marketplaces.
Locked and Crossed Orders	Comments received were split between those supportive of limiting the provisions to protected orders only, and those who disagreed with the proposed approach. Concerns were expressed by some in relation to complexities and the potential for investor confusion. Some believed that a dealer should be prohibited from locking or crossing orders on marketplaces to which they have access.	We continue to believe that instances of locked and crossed orders will be short in duration and are a necessary trade-off to achieve the intended outcomes of the market share threshold (providing for choice to manage inefficiencies and costs). We considered the alternative suggested by some commenters, but are concerned about the increased complexity and costs of compliance monitoring by regulators that would result from such an approach.

Best Execution Obligations and Disclosure	Most commenters were supportive of both the proposed best execution guidance in 23-10CP as well as the proposed best execution disclosure requirements.	We acknowledge the comments received. As discussed, we have postponed finalization of the best execution proposals in order to align the timing with similar proposals by IIROC.
Trading Fee Caps	Most commenters were supportive of the proposed trading fee caps but indicated that the cap proposed for securities priced at or above \$1.00 is too high. A number of commenters indicated that this cap should be lower to better reflect the lower average price of Canadian securities relative to the United States.	As was proposed in the 2014 Notice, we will introduce a \$0.0030 per share cap on active trading fees for securities priced at and above \$1.00, and \$0.0004 per share for securities priced below \$1.00.
	Some commenters were not supportive of introducing trading fee caps on the basis that trading fees are already subject to competitive forces and are approved by regulators.	We acknowledge the concerns expressed by commenters regarding the cap proposed for securities priced at \$1.00 and above. However, our market is highly integrated with the U.S. and we are concerned about the potential negative consequences for the Canadian market if we implement a trading fee cap for U.S. inter- listed securities that is different than the cap in the U.S.
		However, we are proposing a lower active trading fee cap for securities priced at \$1.00 and above that are listed on a Canadian exchange, but not listed on a U.S. exchange (Non-Inter-Listed Securities). This new cap is being published in a separate CSA notice and request for comment.
Prohibition	A number of commenters were supportive of	As noted, we are not

on Payment of Rebates – Pilot Study	regulatory action with respect to the payment of rebates. Some however, suggested that the CSA consider whether rebates are appropriate for certain securities such as those that are less liquid. Many commenters were supportive of the proposed pilot study but cautioned that the study must be carefully designed and appropriate metrics utilized. Others were not supportive of either the pilot study or any action on rebates in general. Certain commenters expressed the view that rebates are an important incentive for liquidity, and that prohibiting rebates will not solve for the issues identified in the 2014 Notice. Some commenters were not supportive of the proposed pilot study and were concerned about the potential outcomes and impacts on the attractiveness of the Canadian market. One commenter questioned whether issuers would be permitted to decline to participate in the study, given the potential impact on liquidity.	proceeding with any action on rebates at this time. Although we are still supportive of a pilot study, we do not believe that meaningful results can be obtained from a study that does not include Inter-Listed Securities. We will continue to liaise with our regulatory counterparts in the U.S and will consider a joint pilot study in the future if an opportunity arises.
Data Fees Methodology	Some commenters were supportive of the proposed methodology, indicating that it is a critical component in addressing some of the issues related to the level of real-time market data fees charged by certain marketplaces. Others were not supportive of the use of the data fee methodology because, in their view, it did not address all the concerns expressed by the users of data. Specifically, one commenter expressed the view that the use of the methodology fails to address the issue of access to a consolidated data feed for all users at a reasonable price. Another commenter indicated that current fees should not be considered as the base for calculating the domestic benchmark as they are too high.	We acknowledge all the comments received regarding the methodology and its application. We continue to be of the view that the data fee methodology is the most appropriate tool to manage some of the existing fee levels. We will continue to monitor the application of the methodology with respect to current and proposed fees and if necessary, will adjust it over time. Additionally, in response to the comments received, we have

	A number of commenters expressed mixed views regarding the methodology and its use. One of these commenters, while supportive of the use of the methodology, pointed out that we should be focusing more on whether the aggregate amount of data fees is fair and reasonable, and less on the redistribution of this amount between marketplaces. Another commenter suggested that the practice of charging for data multiple times for a single user should be eliminated. Another commenter indicated that smaller marketplaces should not be allowed to charge for data. Some commenters also provided their views regarding certain pre- and post-trade metrics. Specifically, concerns were raised with respect to a particular pre-trade metric that would inflate the ranking of a marketplace that displays illiquid securities.	eliminated one pre-trade metric and the ranking model that used this metric. We also acknowledge the comments received in relation to the use of the domestic reference benchmark. It is our intention to engage external assistance in determining the appropriate benchmark. In the meantime, while we will use the domestic benchmark, we will not apply the methodology or the benchmark to support any fee increases by marketplaces until such time as the appropriate benchmark has been established.
Membership and connectivity fees	A number of commenters expressed views on the regulation of membership and connectivity fees. The responses were mixed. The majority of commenters that responded to this question believe that marketplace membership and connectivity fees should be regulated as market participants are required by regulation to connect to protected marketplaces. Several of these commenters explained that this is particularly important to ensure that marketplaces do not institute unwarranted increases of these fees, for example, to mitigate lost revenues due to forthcoming restrictions on other marketplace fees. Other commenters believe that additional regulation is unnecessary in this space because fees are reviewed by regulators currently and because market participants have the ability to connect indirectly to protected marketplaces. Several of these	We will not make regulatory changes in relation to membership and connectivity issues at this time but will continue to monitor such fees to determine if future action is warranted

commenters believe that membership and connectivity fees should be monitored by
regulators.

Торіс	Summary of Comments	CSA Response
General Comments	Many commenters provided their views specifically on the revised Alpha model and approval process that resulted in the implementation of an order processing delay.	The Alpha model was subject to a separate comment period, and the Ontario Securities Commission received and considered all feedback associated with this proposal during the review and approval process.
		Further, as it relates to a 'hybrid' OPR environment where some marketplaces display protected orders while others do not, we received feedback on the same complexities as part of the 2014 proposed OPR amendments (in the context of a market share threshold). The Ontario Securities Commission considered all of these comments in the context of the Alpha proposal and the CSA has considered them in the context of the Proposed Amendments.
	Many commenters expressed a preference for repealing OPR and moving to a best execution model.	We continue to believe that the objectives of OPR are important, and we support maintaining the rule as part of the regulatory framework, in order to promote confidence and provide an incentive to contribute to the price discovery process. Best execution and OPR are complementary rules; one is an obligation to the market as a whole, and the other a duty owed to individual clients. We have discussed the approach of

# Summary of Comments and CSA Responses to 2015 Notice

	Some commenters expressed the belief that regulators should provide enhanced best execution guidance in relation to a 'hybrid' OPR environment – specifically, how an unprotected marketplace should be considered from the perspective of best execution.	moving solely to a best execution model at length, and are not yet satisfied that this approach would result in the best outcome for the entire market. Historically, the approach to best execution has been principles-based, rather than a 'checklist' of factors. A 'hybrid' OPR environment will require additional considerations by dealers when determining whether to access 'unprotected' marketplaces from the perspective of best execution. We also note that we have finalized elements of the 2014 OPR proposals that provide additional guidance for best execution. The CSA along with IIROC, will consider whether further guidance is necessary.
Question 1: Should OPR apply to marketplaces that impose an order processing delay? If so, should it apply to some or all? What factors should be considered?	A number of commenters believed that OPR should not apply to displayed orders on marketplaces that impose an order processing delay, and that any marketplace that imposes a delay should be considered on similar terms. One commenter noted that protecting quotes on 'speedbump' markets would be contrary to the principles OPR was designed to uphold. One commenter was of the view that OPR should continue to apply to marketplaces that impose order processing delays. Some commenters expressed the view that all order processing delays should not be treated equally. Some commenters expressed the view that	Our view continues to be that where a marketplace does not provide for the ability for an immediate execution against displayed volume, that marketplace does not offer "automated trading functionality" as currently defined in NI 23-101. Therefore under the rules, such marketplaces are not displaying "protected orders".
	greater consideration and clarity is required	proposed language in 23-

	regarding what the CSA would consider to be an "intentional order processing delay".	101CP to provide additional clarity.
	A number of commenters indicated that a 'hybrid' OPR model was too complex.	We recognize that additional complexity will result from an environment where some visible marketplaces are displaying protected orders, while others are not. However, complexity has been a trend experienced in many global markets, as participants seek solutions to various challenges and issues, and marketplaces utilize technology to innovate and provide solutions. We are supportive of innovation and believe that these complexities will continue to be managed. We will however, continue to ensure that the principles behind the rule framework are maintained, that rules are applied in a consistent manner, and that negative impacts are addressed appropriately.
Question 2: What are the outcomes and impacts of an environment where not all displayed orders are protected under OPR?	A number of commenters expressed concern regarding the National Best Bid and Offer (NBBO). Specifically, questions were raised about the formulation of the NBBO in terms of what data will be considered, whether separate feeds will exist, and which NBBO will be used to determine the required price improvement for trades with dark orders.	Amendments to IIROC's Universal Market Integrity Rules (UMIR) were approved on September 18, 2015. These amendments revised the definitions of "best bid price" and "best ask price" to limit their determination to orders displayed on a "protected marketplace" (as defined in NI 23-101). Additionally, the Information Processor now offers two different feeds: one which represents the NBBO only
		from marketplaces that display protected orders, and one

Some participants expressed concern about the potential for complexity and confusion due to increased locked and crossed orders, while another commenter believed that occasional locked and crossed markets will not result in major impacts.	representing information from all Canadian marketplaces that display orders (both 'protected' and 'unprotected'). Participants can consume those marketplace feeds that are necessary for their requirements. Given the rapid nature of quoting and trading activity, we believe that instances where orders are locked or crossed will generally be short in duration. To provide for a 'hybrid' OPR environment we have finalized amendments to the locked and crossed provisions proposed in the 2014 Notice, that would limit the prohibition on intentional locking and crossing to protected orders only. Further, we note that the requirements will continue to restrict any further orders from being entered that would intentionally lock or cross with a protected order.
A number of commenters expressed concern regarding the treatment of 'unprotected' marketplaces from the perspective of best execution. Some questioned whether dealers will be able to meet their best execution obligations, absent clear regulatory guidance.	We have finalized proposed amendments to NI 23-101 and 23-101CP that provide additional best execution guidance. We believe these changes will assist dealers in managing their obligations in an environment where not all displayed marketplaces are required to be accessed for purposes of OPR. A determination of whether a marketplace should be accessed for purposes of best execution should be made by an individual dealer as part of

		the broader process of establishing policies and procedures. These considerations are dependent on a variety of factors and we remain committed to a principles-based approach to best execution.
Question 3: What are the expected changes or outcomes for retail dealers and retail clients?	Commenters presented mixed views in relation to the impact on retail clients. Some believed that retail investors will have little ability to judge the quality of fills they will receive and that the Amendments will create a complex environment that will not be understood. Some were concerned that retail investors may experience a reduction in the ability to capture liquidity. Others felt that the existence of 'unprotected' marketplaces will have little impact on retail clients and that dealers will have greater flexibility in accessing various marketplaces in the context of best execution.	We will continue to monitor the impacts of the Amendments in terms of outcomes to all market participants, including retail clients. Retail dealers will continue to have a best execution obligation to their clients, and should evaluate their best execution policies and procedures on an ongoing basis to ensure that any decisions to not access 'unprotected' marketplaces continue to be supportable. For retail clients seeking a better understanding of how their dealer manages their orders, the requirements for enhanced disclosure should assist in this process.
Question 4: Are there implications that have not been addressed that should be considered?	Commenters highlighted similar implications and issues noted in response to previous questions. These include concerns about the determination of the NBBO, impact on the ability to gauge accessible liquidity, impacts on best execution, and impact on market integrity.	Please see responses above. We reiterate that we will continue to monitor the impacts of order processing delays on all market participants and if negative outcomes result, we will take appropriate action.
	One commenter expressed the view that a marketplace could display both protected and unprotected orders and suggested that there are other "systematic delays" that should be	We are of the view that where a marketplace imposes an order processing delay, that marketplace does not provide

treated similarly.	"automated trading functionality". To meet the requirements of the definition, we believe that a marketplace must provide the ability for an immediate execution for all orders. As a result, we do not support the suggestion that where a delay is imposed, a marketplace could display both protected and unprotected orders.
	We have amended the original proposed language in 23- 101CP to provide greater clarity around the consideration and treatment of intentional order processing delays.

# ANNEX F

# **Data Fees Methodology**

The Data Fee Methodology described below will be used to determine each marketplace's relative contribution to pre- and post-trade activities. The scope of the methodology is to determine whether the professional market data fees charged by the marketplaces in Canada reflect each marketplace's share of trading activity.

The methodology consists of three steps:

- 1. Calculation of pre- and post-trade metrics
- 2. Ranking of marketplaces based on the pre- and post-trade metrics calculated in step 1
- 3. Assigning an estimated fee range to each marketplace.

The methodology uses the following notations for the pre- and post-trade metrics and the ranking methods:

Notation	Description
i	A transparent marketplace
m	Total number of transparent marketplaces
j	Securities traded on a transparent marketplace
J	Total securities traded on all transparent marketplaces
t	A Trade executed on a transparent marketplace
n	Total trades executed on a transparent marketplace
Т	Total trades executed on all transparent marketplaces
d	A trading day
D	All trading days for the period under review

# a. <u>Pre-Trade Metrics</u>

1. Percent of Best Bid and Offer  $(BBO)^{11}$  - means the percent of the day for which a marketplace had a quote at the national best bid (BB) or best offer (BO) for security *j*. This metric is scaled to sum to one.

$$\% BBO_{i} = \frac{BBO_{i}}{\sum_{i=1}^{m} BBO_{i}}$$
$$BBO_{i} = \frac{1}{J} \sum_{j=1}^{J} \frac{Seconds \text{ at } BB_{j} + Seconds \text{ at } BO_{j}}{2 * (6.5 * 60 * 60)} * 100$$

<sup>&</sup>lt;sup>11</sup> The time at BBO could be calculated in fractions of a second, given the rapidity of quoting.

This metric rewards marketplaces for being at the BBO for a longer period during the day. This metric is constructed from standard quote data. In order to ensure that the addition of each marketplace sums to one, the individual metrics for each marketplace are summed to come up with a market-wide daily percent at the BBO, and each individual marketplace's percentage is then divided by this total to scale the metric to one.

2. Percent of Best Spread - means the percent of the day that a marketplace was quoting the narrowest spread for security *j*. This metric is scaled to sum to one.

$$\% Spread_{i} = \frac{Spread_{i}}{\sum_{i=1}^{m} Spread_{i}}$$
$$Spread_{i} = \frac{1}{J} \sum_{j=1}^{J} \frac{Seconds \ at \ tightest \ spread_{j}}{6.5 * 60 * 60} * 100$$

This metric tends to reward marketplaces for providing liquidity at both the BB and BO, by establishing the narrowest spread on the market. This metric is also constructed from quote level data. In order to ensure that the addition of each marketplace sums to one, the individual metrics for each marketplace are summed to come up with a market-wide daily percent at the narrowest spread, and each individual marketplace's percentage is then divided by this total to scale the metric to one.

**3. \$Time(value)** - means the percent of quoted-time-dollar-volume for a marketplace, out of the total time-dollar-volume for the entire market for the period, when only the BB and BO are considered. Each stock is weighted by the value traded in the period of consideration, as described in the weighting "w" below.

$$\begin{aligned} \$ Time(value)_{i} &= \frac{Time(v)_{i}}{\sum_{i=1}^{m} Time(v)_{i}} Time(v)_{i} \\ &= \frac{\sum_{j=1}^{J} [Price_{j} * Volume_{j} * seconds at BB + Price_{j} * Volume_{j} * seconds at BO] * w_{j}}{\sum_{j=1}^{J} \sum_{i=1}^{m} (Price_{j} * Volume_{j} * seconds at BB + Price_{j} * Volume_{j} * seconds at BO) * w_{j}} \\ &= 100 \end{aligned}$$

$$w_{j} = \frac{\$Volume_{t,j}}{\sum_{t=1}^{T} \sum_{i=1}^{J} \$Volume_{t,j}}$$

The use of the value weighting places more emphasis on those stocks that trade heavily and less emphasis on stocks that do not trade frequently. At the extreme, a stock that does not trade at all will not be allocated any weight under this metric.

# b. Post-Trade Metrics

**1. Percent of each marketplace's volume** - means the volume traded on each marketplace divided by the total volume traded on all marketplaces in the period.

$$\%Volume_i = \frac{Volume_i}{\sum_{i=1}^m Volume_i} * 100$$

This metric rewards traded volume and tends to favour those marketplaces that trade in relatively low-priced shares, as it considers only the number of shares traded, not their value. In an extreme scenario, if a marketplace traded only low-priced stocks, this metric would inflate their overall share of the entire market.

2. Percent of each marketplace's number of trades - means the number of trades executed on each marketplace divided by the total number of trades on all marketplaces in the period.

$$\%Number_i = \frac{Number_i}{\sum_{i=1}^{m} Number_i} * 100$$

This metric rewards those marketplaces that have a larger number of trades. This metric could be manipulated by encouraging traders to break their orders up into smaller pieces. If this were done, neither the volume nor the dollar volume traded would change, but the number of trades would increase significantly.

**3. Percent of each marketplace's dollar volume (value)** - means the dollar volume traded on each marketplace divided by the total dollar volume traded on all marketplaces in the period. Dollar volume is the product of the price and volume of each trade.

% 
$$Volume_i = \frac{Volume_i}{\sum_{i=1}^m Volume_i} * 100$$

This metric takes the value of the transactions into account. This tends to avoid the biases that may be present in the volume metric. However, due to the requirement that crosses matched by a dealer be reported to a marketplace, it is possible that a marketplace being measured on this metric could provide incentives (such as trading rebates) to dealers to ensure that crosses are reported on their marketplace. In this way, the marketplace would have a much larger share of dollar volume without necessarily contributing to pre-trade price discovery.

**4. Percent of square-root dollar volume for each trade** - means the square root of the \$Volume of each trade *t* executed on each marketplace divided by the sum of the square-root of the \$Volume traded on all marketplaces in the period.

$$\%\sqrt{\$Volume_i} = \frac{\sum_{t=1}^n \sqrt{\$Volume_{i,t}}}{\sum_{t=1}^n \sum_{i=1}^m \sqrt{\$Volume_{i,t}}} * 100$$

The square-root of dollar volume is individually constructed for each transaction. This metric is not widely published, but it is easily constructed from trade reports. This metric reduces the importance of larger trades in relation to smaller trades. This can help alleviate the problem of very large crosses inflating a marketplace's contribution to price discovery. This metric has the potential disadvantage that trades in low-priced stocks (on the order of \$1 to \$2) will not be reduced at all, and will consequently be disproportionately represented. If a marketplace were to trade very frequently at these very low dollar values, their contribution to price discovery would be inflated by this metric.

5. Scope of trading on each marketplace - means the average over the period of the number of symbols with greater than 1 traded on each marketplace on day *d*, divided by the number of symbols traded on all marketplaces for that day.

$$Scope_{i} = \frac{1}{D} \sum_{d=1}^{D} \frac{Number \ of \ symbols \ traded_{i,d}}{MAX[Number \ of \ symbols \ traded_{i,d}]}$$

Scope of trading provides a metric that measures the number of symbols a marketplace trades. This metric, when used in combination with other post-trade metrics, has the disadvantage of "double penalizing" marketplaces for not trading all securities. By construction, scope of trading will be very high for exchanges (such as the TSX) and will be lower for newer marketplaces that have yet to gain market share in less liquid stocks. While it does measure the "activity" of marketplaces, a marketplace that only trades in half of the total listed symbols is, by definition, penalized for not trading all of those symbols. Thus, if scope is used by itself, it can be a valuable indicator of the activity levels of marketplaces, but if it is applied in conjunction with other metrics, it may disproportionately favour existing exchanges and large ATSs.

The downside of this metric is that if a marketplace wanted to achieve a scope as close as possible to one (i.e. all listed securities would be trading on this marketplace), marketplace participants could be rewarded (through credits or discounts at market open) for becoming the "first" participant of the day in any given security. In this way, marketplaces could ensure at least one trade in every security without providing any meaningful liquidity or price discovery.

### c. Ranking Models

In order to rank each marketplace's contribution to price discovery we constructed two models from the pre- and post-trade metrics. While each model is constructed placing equal importance on the pre- and post-trade metrics, this was an arbitrary decision.

1. SIP Value – is based on the revenue distribution model used by the U.S. SIP.

$$\left[\frac{\sqrt[6]{Volume_i} + \sqrt[6]{Number_i}}{2}\right] * 0.5 + Time(value)_i * 0.5$$

This model incorporates the metrics used by the U.S. SIP to distribute revenue amongst participating marketplaces. The post-trade metrics used are equally weighted, and are composed of each marketplace's share of square root dollar volume and number of trades. Both of these post-trade metrics together are assigned a weighting of 50% of the value of the model.

The pre-trade metric used is the value weighted percent of quoted dollar-time. This is also given a 50% weighting in the final model. The weighting of this model by the value traded in each security provides a greater emphasis on those stocks that are heavily traded, rewarding marketplaces more for providing liquidity where the majority is consumed.

2. Model 3 - differs significantly from the previous model. For the post-trade element, this model considers each marketplace's share of traded volume, share of trades and share of dollar-volume. These three elements are given equal weighting in this index. The pre-trade metrics considered are the percent of the day spent at the best spread and the percent of the day spent at the BBO. Each of these two pre-trade elements is equally weighted. The resulting pre- and post- trade metrics are then equally weighted to come up with the final index.

 $\left[\frac{\%Volume_i+\%Number_i+\%Volume_i}{3}\right]*0.5+\left[\frac{\%Spread_i+\%BBO_i}{2}\right]*0.5$ 

# d. Assigning an estimated fee or fee range

After calculating these ranking methods, we would use them to assess whether a marketplace's existing (or proposed) fee is related to its share of trading activity. We use the domestic reference that takes the data fees charged by each marketplace and aggregates them into a single "pool". The result is then considered to be the appropriate fee for the Canadian market, and this result is then re-distributed, based on the two ranking models, giving us four estimated fees for each marketplace.

# ANNEX G

## **Local Matters**

In Ontario, the amendments to National Instrument 23-101 and other required materials were delivered to the Minister of Finance on April 5, 2016. The Minister may approve or reject the amendments or return them for further consideration. If the Minister approves the amendments or does not take any further action by June 20, 2016, the amendments will come into force on July 6, 2016, except as related to the market share threshold. The amendments related to the market share threshold will come into force on October 1, 2016.